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WPф ɳˉʽˁˇʽ˄˖˄ʾʰ ˁʰʽ ʵʽʱʵˇˋʹ 

DΦфΦм ɹˋˍˇˋʶ˂ʾʵʰ ˍˇˎ ʷˊʴˇˎΦ 

ɲʹ˃ʽˇˎˊʴʺʻʹˁʶ ʹ ʽˋˍˇˋʶ˂ʾʵʰ ˍˇˎ ʷˊʴˇˎ DENOMINATE ώɾмуϐ 

 

 

ʂˉˇˎ ˁʰˍʰ˔˖ˊʺʻʹˁʰ˄ ˈ˂ʶˌ ˇʽ ˋʹ˃ʰ˄ˍʽˁʷˌ ˉ˂ʹˊˇ˒ˇˊʾʶˌ ˍˇˎ ʷˊʴˇˎ ˈˉ˖ˌΥ 

¶ ʹ ˉʶˊʽʴˊʰ˒ʺ ˍˇˎ ʷˊʴˇˎ 

¶ ˇʽ ˋˍˈ˔ˇʽ ˁʰʻ˗ˌ ˁʰʽ  

¶ ʹ ˃ʶʻˇʵˇ˂ˇʴʾʰ ˉˇˎ ʰˁˇ˂ˇˎʻʺʻʹˁʶ 



 

 

 



 

DфΦн ɲˊʰˋˍʹˊʽˈˍʹˍʶˌ ˁʰʽ ʵʽʱʵˇˋʹ ˍ˖˄ ʰˉˇˍʶ˂ʶˋ˃ʱˍ˖˄ 

ɶ ̌ ˃ʱʵʰ ˍˇˎ DENOMINATE h ˉʰˊˍʾʸʶˍʰʽ ʰˉˈ ˍʰ ʰˁˈ˂ˇˎʻʰ ʶˊʶˎ˄ʹˍʽˁʱ ˃ʷ˂ʹ ˈˉˇˎ ʲˇʺʻʹˋʰ˄ ʴʽʰ ˍʹ˄ 
ˇ˂ˇˁ˂ʺˊ˖ˋʹ ʰˎˍˇˏ ˍˇˎ ʷˊʴˇˎΥ 

ɼʰʻΦ ɱʶˊʱˋʽ˃ˇˌ ɽˎ˃ˉʶˊʱˍˇˌ όʶˉʽˋˍʹ˃ˇ˄ʽˁˈˌ ˎˉʶˏʻˎ˄ˇˌύ 

ɲˊΦ ɼ˖˄ˋˍʰ˄ˍʾ˄ʰ ʃʰˉʰʵˇˉˇˏ˂ˇˎ όɳɲɹʃύ 

ɲˊΦ ɮˋʹ˃ʾ˄ʰ ʆˊʶ˃ˇˏ˂ʹ ό˃ʶˍʰʵʽʵʱˁˍ˖ˊύ 

ɲˊΦ ɱʶ˗ˊʴʽˇˌ ɾʱˊʽˇˌ ɽˏˍˊʰˌ ό˃ʶˍʰʵʽʵʱˁˍ˖ˊύ 

ɲˊΦ ɸʶˈ˒ʽ˂ˇˌ ɼʰ˃ˉʶˊʾʵʹˌ ό˃ʶˍʰʵʽʵʱˁˍ˖ˊύ 

ɮ˔ʽ˂˂ʷʰˌ ɵʰˊˁʰ˂ʾˇˎ όˎˉˇ˕ʺ˒ʽˇˌ ʵʽʵʱˁˍ˖ˊύ 

ɱʶˊʱˋʽ˃ˇˌ ɼʰ˄ʷ˂˂ˇˌ όˎˉˇ˕ʺ˒ʽˇˌ ʵʽʵʱˁˍ˖ˊύ 

ʅˍʰ ˉ˂ʰʾˋʽʰ ˍˇˎ ʷˊʴˇˎ DENOMINATE ̌ ˂ˇˁ˂ʹˊ˗ʻʹˁʰ˄ у ʵʽˉ˂˖˃ʰˍʽˁʷˌ όˉʰˊʰˁʱˍ˖ ˉʾ˄ʰˁʰˌύΣ ʶ˄˗ ʶʾ˄ʰʽ ˎˉˇ 
ˇ˂ˇˁ˂ʺˊ˖ˋʹ ˍʰ ʵʽʵʰˁˍˇˊʽˁʱ ˍ˖˄ ɮΦ ɵʰˊˁʰ˂ʾˇˎ ˁʰʽ ɱΦ ɼʰ˄ʷ˂˂ˇˌΣ ˋˍʹ˄ ʅ˔ˇ˂ʺ ʋʹ˃ʽˁ˗˄ ɾʹ˔ʰ˄ʽˁ˗˄ ɳɾʃ ˋˍˇ 
ʶˊʴʰˋˍʺˊʽˇ ʁˊʴʰ˄ʽˁʺˌ ʋʹ˃ʽˁʺˌ ʆʶ˔˄ˇ˂ˇʴʾʰˌΦ 

 

 

 



ʂ˄ˇ˃ʰ ɳˉ˗˄ˎ˃ˇ ʆʾˍ˂ˇˌ 

ɮɿʆɹɱʁɿɶ ʅʆɮɾɳɽʁʇ ɮ˅ʽˇˉˇʾʹˋʹ ˎʴˊˇˏ ˁ˂ʱˋ˃ʰˍˇˌ 
ʰˉˇʲ˂ʺˍ˖˄ ˍˊˇ˒ʾ˃˖˄ ʴʽʰ ˉʰˊʰʴ˖ʴʺ 
ʰʷˊʽ˖˄ ʲʽˇˁʰˎˋʾ˃˖˄ ˋʶ ʰ˄ˍʽʵˊʰˋˍʺˊʰ 
CSTR 

ɳʇɮɱɱɳɽɹɮ ʃʄɹʊʆɶ ʅˎ˄-ʶˉʶ˅ʶˊʴʰˋʾʰ ʰˋˍʽˁ˗˄ ˎʴˊ˗˄ 
ʰˉˇʲ˂ʺˍ˖˄ ˋʶ ʰ˄ˍʽʵˊʰˋˍʺˊʰ SBR 
(Sequencing Batch Reactor ς 
ʰ˄ˍʽʵˊʰˋˍʺˊʰˌ ʵʽʰ˂ʶʾˉˇˎˋʰˌ 
˂ʶʽˍˇˎˊʴʾʰˌύ 

ɮɿɿɮ ɾʍɼʁʇ ɳˉʶ˅ʶˊʴʰˋʾʰ ʰˋˍʽˁ˗˄ ˎʴˊ˗˄ 
ʰˉˇʲ˂ʺˍ˖˄ ˋʶ ˍʰ˔ˏˊˊˎʻ˃ˇ 
ʲʽˇʰ˄ˍʽʵˊʰˋˍʺˊʰ όPABRύΣ ˃ʶ 
˂ʶʽˍˇˎˊʴʾʰ όABR) 

ɹʍɮɿɿɮ ς 
ɳɽɳɿɶ 

ɼɹʁʇɼɶ ɳˉʶ˅ʶˊʴʰˋʾʰ ˎʴˊ˗˄ ʰˉˇʲ˂ʺˍ˖˄ ˋʶ 
ʰ˄ˍʽʵˊʰˋˍʺˊʰ SBR (Sequencing Batch 
Reactor - h ˄ˍʽʵˊʰˋˍʺˊʰˌ 
ʵʽʰ˂ʶʾˉˇˎˋʰˌ ˂ʶʽˍˇˎˊʴʾʰˌύ 

ɲɶɾɶʆʄɹʁʅ ɾɹʋʁʃʁʇɽʁʅ ʅˍʰʻʶˊˇˉˇʾʹˋʹ ʰˉˇʲ˂ʺˍ˖˄ ʰˉˈ 
˒˖ˍˇʲˇ˂ˍʰʿˁʱ ˉʱ˄ʶ˂ 

ɮɿʆʍɿɹʁʅ ɼʁɿɲʇɽɶʅ ʅˎ˄ʶˉʶ˅ʶˊʴʰˋʾʰ ˎʴˊˇˏ ˁ˂ʱˋ˃ʰˍˇˌ 
ˍˊˇ˒ʽˁ˗˄ ʰˉˇʲ˂ʺˍ˖˄ ˃ʶ 
˂ˎ˃ʰˍˇ˂ʱˋˉʹ ʴʽʰ ˉʰˊʰʴ˖ʴʺ ʰʷˊʽ˖˄ 
ʲʽˇˁʰˎˋʾ˃˖˄ ˋʶ ʰ˄ˍʽʵˊʰˋˍʺˊʰ CSTR 

ɳˎ˒ˊˇˋˏ˄ʹ ɼˇˎˍˊˇˎ˃ʱ˄ˇˎ ɾʶ˂ʷˍʹ ʶ˄ʶˊʴʶʽʰˁʺˌ ʰ˅ʽˇˉˇʾʹˋʹˌ ˍˇˎ 
ˋˎ˃ˉˎˁ˄˗˃ʰˍˇˌ ̱ ʹˌ ˅ʺˊʰ˄ˋʹˌ 
ʸˎ˃˗ˋʽ˃˖˄ ʰˋˍʽˁ˗˄ ʰˉˇʲ˂ʺˍ˖˄ 

ɳˎʱʴʴʶ˂ˇˌ 
ɾʱˊʽˇˌ  

ɼʰˋˍˊʾ˄ʹˌ ɾʶ˂ʷˍʹ ɲʽʰ˒ˇˊʶˍʽˁ˗˄ ʇ˂ʽˁ˗˄ 
ɼʰˍʰˋˁʶˎʺˌ ˋʶ ɾʽˁˊˇʲʽʰˁʺ ɼˎ˕ʶ˂ʾʵʰ 
ɼʰˎˋʾ˃ˇˎ 

 

ɶ ʵʽʱʵˇˋʹ ˍ˖˄ ʰˉˇˍʶ˂ʶˋ˃ʱˍ˖˄ ˍˇˎ ʷˊʴˇˎ DENOMINATE ˉˊʰʴ˃ʰˍˇˉˇʽʺʻʹˁʶ ˃ʶ н ʵʹ˃ˇˋʽʶˏˋʶʽˌ ˋʶ ʵʽʶʻ˄ʺ 
ˉʶˊʽˇʵʽˁʱ ˃ʶ ˁˊʽˍʷˌ ˁʰʽ ф ʰ˄ʰˁˇʽ˄˗ˋʶʽˌ ˋʶ ʵʽʶʻ˄ʺ ˁʰʽ ʶʻ˄ʽˁʱ ˋˎ˄ʷʵˊʽʰΣ ˈˉ˖ˌ ˉʰˊˇˎˋʽʱʸˇ˄ˍʰʽ ˉʰˊʰˁʱˍ˖Υ 

ɲʹ˃ˇˋʽʶˏˋʶʽˌ ˋʶ ʵʽʶʻ˄ʺ ˉʶˊʽˇʵʽˁʱ ˃ʶ ˁˊʽˍʷˌ 

1. G.M. Lytras, E. Koutroumanou, and G., Lyberatos,  Anaerobic co-digestion of condensate produced 
from drying of Household Food Waste and Waste Activated Sludge, Journal of Environmental Chemical 
Engineering, Volume 8, Issue 4, 2020,  DOI: 10.1016/j.jece.2020.103947. 

2. Zarkaliou A., Kougias C., Mokou A., Papadopoulou K*. Lyberatos G. Anaerobic Digestion of Synthetic 
Municipal Wastewater (MWW) in a Periodic Anaerobic Baffled Reactor (PABR): Assessment of COD 



Removal and Biogas Production, Applied Sciences (2022), 12 (24), 
https://doi.org/10.3390/app122413037.  

 

ɮ˄ʰˁˇʽ˄˗ˋʶʽˌ ˋʶ ʵʽʶʻ˄ʺ ˋˎ˄ʷʵˊʽʰ 

3. G. Kanellos, A. Tremouli, G. Lytras, A. Kondylis, G. Lyberatos, Co-digestion of the liquid fraction of food 
waste with waste activated sludge, 8th International conference of engineering for waste and biomass 
valorization, WasteEng2020. 

4. Lyberatos G., keynote speaker, Integrated management of municipal wastewater with source 
collected biowaste"International Conference "Protection & Restoration of the Environment - tw9ʋ±ɹϦΣ 
Kalamata, July 5-у нлннΣ ɳƭƛǘŜ wŜǎƻǊǘ IƻǘŜƭ 

5. A. Zarkaliou, C. Kougias, K. Papadopoulou, G. Lyberatos, Anaerobic Digestion of Municipal Wastewater 
(MWW) in a Periodic Anaerobic Baffled Reactor (PABR), 17th International Conference on 
Environmental Science and Technology Athens, Greece, 1 to 4, CEST2021. 

6. T. Kamperidis, P. Pandis, E. Vlachaki, A. Tremouli, G. Lyberatos, Condensate originating from household 
fermentable waste as a substrate for microbial fuel cells, 17th International Conference on 
Environmental Science and Technology Athens, Greece, 1 to 4 September 2021, CEST2021. 

7. Zarkaliou A., Papadopoulou K.*, Mokou A., Kiouki E., Lyberatos G., Development of a novel concept 
for interrated management of municipal wastewater and biowaste. 9th International Conference on 
Engineering for Waste and Biomass Valorisation WasteEng2022, Copenhagen (Denmark), June 27-30, 
2022 

8. Zarkaliou A. Prifti E., Kokkas T., Papadopoulou K.*, Lyberatos G., Development of an innovative 
scheme for the simultaneous treatment of municipal wastewater and the condensate from drying 
food waste, 18th International Conference on Environmental Science and Technology. CEST2023, 30 
August to 2 September 2023, Athens, Greece. 

9. Lyberatos G., keynote presentation Combined management of food waste and municipal wastewater, 
18th International Conference on Environmental Science and Technology. CEST2023, 30 August to 2 
September 2023, Athens, Greece. 
 

ɮ˄ʰˁˇʽ˄˗ˋʶʽˌ ˋʶ ʶʻ˄ʽˁʱ ˋˎ˄ʷʵˊʽʰ 

млΦ ɵʰˊˁʰ˂ʾˇˎ ɮΣ ʃʰˉʰʵˇˉˇˏ˂ˇˎ ɼϝΣ ɽˎ˃ˉʶˊʱˍˇˌ ɱΣ ɮ˄ʱˉˍˎ˅ʹ ɼʰʽ˄ˇˍˈ˃ˇˎ ʃˊˇˋʷʴʴʽˋʹˌ ʴʽʰ ˍʹ˄ 
ʁ˂ˇˁ˂ʹˊ˖˃ʷ˄ʹ ɲʽʰ˔ʶʾˊʽˋʹ ˍ˖˄ ɮˋˍʽˁ˗˄ ʇʴˊ˗˄ ɮˉˇʲ˂ʺˍ˖˄ ˁʰʽ ɰʽˇʰˉˇʲ˂ʺˍ˖˄Σ моƻ ʃʰ˄ʶ˂˂ʺ˄ʽˇ 
ɳˉʽˋˍʹ˃ˇ˄ʽˁˈ ʅˎ˄ʷʵˊʽˇ ʋʹ˃ʽˁʺˌ ɾʹ˔ʰ˄ʽˁʺˌ ʃʱˍˊʰΣ н-п ɹˇˎ˄ʾˇˎ нлнн 

ммΦ ɮΦ ɵʰˊˁʰ˂ʾˇˎΣ ɼΦ ʃʰˉʰʵˇˉˇˏ˂ˇˎϝΣ ɱΦ ɽˎ˃ˉʶˊʱˍˇˌ ʁ˂ˇˁ˂ʹˊ˖˃ʷ˄ʹ ʵʽʰ˔ʶʾˊʽˋʹ ˍ˖˄ ʲʽˇʰˉˇʲ˂ʺˍ˖˄ ˁʰʽ 
ʰˋˍʽˁ˗˄ ˎʴˊ˗˄ ʰˉˇʲ˂ʺˍ˖˄ ʴʽʰ ˉʰˊʰʴ˖ʴʺ ʲʽˇʰʶˊʾˇˎΣ ɳɽɽɶɿɹɼɶ ɳʆɮɹʄɳɹɮ ɲɹɮʋɳɹʄɹʅɶʅ ʅʆɳʄɳʍɿ 
ɮʃʁɰɽɶʆʍɿ ˋʶ ˋˎ˄ʶˊʴʰˋʾʰ ˃ʶ ˍʹ˄  LƴǘŜǊƴŀǘƛƻƴŀƭ {ƻƭƛŘ ²ŀǎǘŜ !ǎǎƻŎƛŀǘƛƻƴ όL{²!ύΣ ζɳˉʽˍʱ˔ˎ˄ˋʹ ˍʹˌ 
ɾʶˍʱʲʰˋʹˌ ˋˍʹ˄  ɼˎˁ˂ʽˁʺ ʁʽˁˇ˄ˇ˃ʾʰ - ɳˎˁʰʽˊʾʶˌ ϧ ɼʾ˄ʵˎ˄ˇʽηΣ нуτнф ʅʶˉˍʶ˃ʲˊʾˇˎ нлнн 
όʃˇ˂ˎˍʶ˔˄ʶʾˇΣ ʃʰˍʹˋʾ˖˄ύ 
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Highlights 

ω {ǳƛǘŀōƭŜ ƳƛȄƛƴƎ ƻŦ ŘƛŦŦŜǊŜƴǘ ǿŀǎǘŜ ǎǘǊŜŀƳǎ ǘƻ ŀŎƘƛŜǾŜ ƘƛƎƘ ǇŜǊŦƻǊƳƛƴƎ !ƴŀŜǊƻōƛŎ 5ƛƎŜǎǘƛƻƴΦ 

ω /{¢w ŀŎƘƛŜǾŜǎ Ҕфл҈ /h5 ǊŜƳƻǾŀƭ ŦƻǊ ƳƛȄŜŘ ŦŜŜŘ ƻŦ ǿŀǎǘŜǎΦ 

ω .ƛƻƎŀǎ productivity >20L/d, with a methane content of up to 80%. 

Purpose 

Anaerobic Digestion (A.D.) is the biological process during which complex organic compounds of wastes are 
decomposed in the absence of oxygen, by anaerobic microorganisms. Through the intricate sequence of 
actions of these microorganisms (disintegration, hydrolysis, acidogenesis, acetogenesis, methanogenesis), the 
organic substrates are converted into renewable energy, in the form of biogas, while the residue meets the 
specifications for disposal in the soil or conversion into useful by-products [1]. This study aims to suggest a 
novel approach for simultaneous treatment of municipal waste sludge and the liquid fraction of food waste, 
in order to render the A.D. process more stable, increase the biogas yield and treat the mixed waste stream 
more effectively. 

Materials and methods 

Anaerobic Sludge obtained from the Municipal Wastewater Treatment Plant of Attica, Greece was used as 
inoculum for the A.D. reactor during start-up. The substrates to be co-digested were Waste Activated Sludge 
(W.A.S.), obtained from the same treatment plant, and the liquid fraction of household food waste. The food 
waste undergoes a rapid shredding, drying and condensation process and the liquid fraction (condensate), rich 
in organic load, is mixed with W.A.S. and used as substrate to be fed in the bioreactor. The digestion process 
of the mixed substrates takes place in a 100 L anaerobic bioreactor in continuous operation mode. The CSTR 
is kept at a constant hydraulic retention time (HRT) of 20 days, constant temperature at mesophilic conditions 
ƻŦ орх/ ŀƴŘ ƛǎ ƳƻƴƛǘƻǊŜŘ ǘƘǊƻǳƎƘ ŦǊŜǉǳŜƴǘ ƻōǎŜǊǾŀǘƛƻƴ ƻŦ ǾŀǊƛƻǳǎ ƳŜŀǎǳǊŜƳŜƴǘǎ ŀŎŎƻǊŘƛƴƎ ǘƻ {ǘŀƴŘŀǊŘ 
Methods. The Analytical methods used for characterizing the mixed feedstock and digested outlet of the 
bioreactor include pH, alkalinity, solids, total and soluble Chemical Oxygen Demand (COD) measurements, 



Total Organic Carbon (TOC) and Total Nitrogen (TN), dissolved and Total Kjeldahl Nitrogen (TKN), Volatile Fatty 
Acids and alcohols [2]. The biogas production is measured through an oil displacement technique and the 
methane content is quantified using a GC-TCD. 

Results and discussion 

Condensate samples are extracted every hour of the pretreatment process in order to analyze the temporal 
distribution of the waste characteristics. Figure 1a depicts the tCOD of condensate extracted in relation to its 
moisture content, while Figure 1b shows the TOC and TN of the samples. These results are utilized for the 
determination of the mixed feedstock characteristics. During startup, the inoculum included a mixture of 
Anaerobic Sludge, W.A.S. and Condensate in ratios 1:17:2 respectively, based on previous studies [3]. 

 

Figure 1: The condensate effluent characteristics versus time showing a) its dry weight (right y-axis) and tCOD 
(left y-axis) content and b) its TOC (right yaxis) and TN (left y-axis) load. 

Following the start-up in batch mode, the bioreactor was shifted to continuous mode. The influent was 
maintained constant at 5 L of mixed W.A.S. and condensate in 2:1 ratio, based on the annual global Municipal 
Solid Waste generation rate [4]. Figure 2 shows the continuous operation sCOD at the inlet and outlet of the 
bioreactor, indicating that the organic matter COD removal exceeds 90% . 

 

Figure 2: The sCOD of the CSTR inlet and outlet versus operation time. 

As a result of the digestion of the mixed feed, biogas is produced. Figure 3 depicts the total biogas productivity, 
as well as its methane content. 

 



Figure 3: The CSTR daily biogas productivity (left y-axis) and its methane content (right y-axis) versus operation 
time. 

Conclusions and perspectives 

Most of the organic matter of the mixed feed is digested. The COD removal was over 90%. The steady state 
operation yielded over 20L biogas per day with a methane content of 80%. Future perspectives include the 
optimization of the CSTR operation, in terms of Organic Loading Rate, search for optimal C:N ratio for digestion 
of the mixed feedstock, HRT, biogas productivity and methane content within the biogas. In addition, 
supplementary BioMethane Potential measurements will be performed in smaller scale bioreactors for 
optimization of the abovementioned parameters. 
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Abstract 

The scope of this work is to study the treatment of municipal wastewater in a Periodic Anaerobic Baffled 
Reactor (PABR). PABR is an innovative, high-rate bioreactor, designed to operate under high organic loadings. 
Apart from the Hydraulic Retention Time (HRT) an important operational parameter is the Switching Period 
(T). The current research work aims to study the impact of the operational parameters variation (HRT and T) 
in the biogas and biomethane productivity. 

Six distinct experimental phases were conducted, during which the operational parameters of the PABR were 
consecutively modified: HRT varied from 10 to 1 day, T between 2.5  and 0.25 days while the OLR remained 
constant at values near 1 gsCOD/Lbioreactor*d. The maximum CH4 productivity was 26.5 LCH4/d corresponding to 
the operation under HRT=1d, OLR=0.89 gsCOD/Lbioreactor*d and T=0.25 days. Conclusively, the PABR is a high-rate 
AD system, capable of treating MWW under extreme operational conditions. 

 

Keywords: Anaerobic Digestion; Bioreactor; High-rate; Methane; Municipal Wastewater; PABR 

1. Introduction 

Currently, the benchmark approach to municipal wastewater (MWW) management consists of sewer 
collection, treatment in a facility aiming at removal of suspended solids through primary sedimentation, 
biological oxidation of organic matter, biological nutrient (N and P) removal and disposal of the clarified 
effluent following disinfection by chlorination. The process generates a mixture of primary and excess 
secondary sludge which are typically mixed, stabilized by anaerobic digestion and dewatered before disposal 
(Chan et al., 2009). The key operating costs lie in the aeration and in sludge (biosolids) management (Mohan 
et al., 2008).  

Alternative approaches that would reduce the energy requirements have been contemplated in the recent 
years. Indeed, it is possible to produce energy from the dissolved organic matter in wastewater, rather than 
consuming energy for aeration, followed by a partial recovery through anaerobic digestion of the biosolids. 
Among them, direct anaerobic digestion (AD) of the wastewater has been examined (Sosnowski et al., 2003). 

The PABR was designed by Skiadas and Lyberatos (1998).  It is an innovative high-rate anaerobic digestion 
system capable of anaerobically processing high organic-loaded feedstocks at low HRTs. As shown in Figure 1 
it consists of two concentric cylinders. The space between the two cylinders is divided into four compartments, 
each one of which is further divided into two sections, one downflow and one upflow, thus resembling a simple 
ABR, only arranged in a circular structure. However, an important property of the specific bioreactor is the 
ability to periodically change the inflow and (outflow) compartment.  

The time required for all the compartments to act as feeding compartments is the switching period T. This 
specific operational parameter gives the bioreactor the element of operational flexibility: when T is high, the 
bioreactors operation is similar to an ABR, while when it is low the operation approaches the one of an Upflow 
Anaerobic Sludge Blanket (UASB). 

 

 

 



 

Figure 1. PABR experimental setup 

 

2. Materials and Methods 

A pilot-scale 77-L active volume PABR was utilized for the AD experiments. The system was fed with a synthetic 
municipal wastewater, operating under mesophilic conditions 35oC, for 147 consecutive days.  

The synthetic wastewater used for the PABR in the present study consists of: 10 to 1.0 g/L glucose, 0.306 to 
0.0285 g/L NH4Cl (regarding the experimental phase),  0.08 g/L CH3COONa, 0.044 g/L KH2PO4, 0.0275 g/L 
MgSO4*7H2O, 0.0025 g/L CaCl2, 0.004 g/L KCl, 0.125 g/L NaHCO3, 1.875 mg/L FeCL3*6H2O, 0.1875 mg/L H3BO3, 
0.225 mg/L KI, 0.15 mg/L MnSO4, 0.0275 mg/L ZnSO4*7H2O, 0.0375 mg/L CuSO4*5H2O and 12.5 mg/L EDTA 
(Shuli Liu et al., 2020). Every experimental phase had different concentration of glucose and ammonium 
chloride so that the OLR would be kept constant while reducing the HRT. The concentrations of the above 
mentioned substances, were chosen by calculating the C/N ratio that would occur so that it remained constant 
near the value of 50. 

The scope of the experimental process was to evaluate the efficiency of the PABR under different conditions 
in terms of organic load reduction and biogas and biomethane productivity. Municipal wastewaters tend to 
have COD concentration lower than 1000mg/L, so by reducing the organic load of the feed mixture we tried 
to approach that value as much as possible.  

Therefore, the bioreactor operated under various HRTs and T while the organic loading rate was kept at values 
near 1 gsCOD/L*d (as outlined in Table 1). In all cases, the ratio of HRT/T was kept constant and equal to 1. 

 

 

 

 

 

Table 1. Operational Parameters, 6 phases, PABR 



Operational 
Parameters 

Experimental Phases 

Phases 1 2 3 4 5 6 

Operation Duration 
(days) 

21 23 20 10 69 4 

HRT (days) 10 6 4 3 2 1 

T switching period 
(days) 

2.5 1.5 1 0.75 0.5 0.25 

Six distinct experimental phases were carried out as shown in Table 1. Throughout the experimental process 
pH, total alkalinity, Total Suspended Solids (TSS), Volatile Suspended Solids (VSS), total and soluble Chemical 
Oxygen Demand (tCOD, sCOD), Volatile Fatty Acids (VFAs) TOC, TN and TKN (data not shown), biogas 
production and methane content were monitored in regular intervals, to assess the efficiency of the process. 
TSS, VSS, tCOD, sCOD and alkalinity were measured according to Standard Methods (APHA, 1995), VFAs were 
measured using a gas chromatograph (SHIMADZU GC-2010 plus), while a GC-TCD (SHIMADZU GC-2014) was 
used for the measurement of the methane content in the generated biogas. Moreover TOC-L Shimadzu was 
used for the measurement of total organic carbon and total nitrogen. 

3. Materials and Methods 

The overall efficiency of the PABR throughout the experimental phases is presented in Table 2: 

 

Table 2. Experimental results, 6 phases, PABR 

   Experimental Phases 

Phases 1 2 3 4 5 6 

OLR (gsCOD/L*d) 0.91 0.96 0.95 0.90 0.85 0.89 

tCOD removal (%) 79.5 83.7 86.9 89.6 85.3 64.9 

Biogas productivity (L/d) 25.6 33.8 37.0 35.6 32.3 44.3 

CH4 productivity (L/d) 6.9 15.2 18.7 18.5 21.1 26.5 

From table 2 it is apparent that HRT reduction from 10 to 1 days significantly affected the biogas and methane 
productivity, as well as the tCOD removal achieved in the PABR respectively. 

 



 

Figure 2. Biogas and biomethane average production throughout the six experimental phases. 

 

Figure 3. COD concentration in the PABR compartments, Feed and Effluent at every operational experimental 
phase. 

As shown in Figure 2 maximum biogas production observed during the 6th experimental phase, reaching 
44.29, while the biomethane production reached 26.5 L/d. As shown in Figure 3 the PABR reactor managed to 
reduce the effluent COD at every experimental phase that operated. 

4. Conclusion 

In this paper, we evaluated the efficiency of a PABR for the treatment of a synthetic municipal wastewater 
operating under different conditions and assessed biogas and biomethane productivity. 

It was shown that the PABR can efficiently operate under HRTs as low as 1 day. Specifically, maximum biogas 
production was observed in the experimental phase when the basic operational parameters were HRT: 1 d 
and the OLR: 0.89 gsCOD/L*d. In those phases biogas production reached 44.3 L/d, while the biomethane 
production reached 26.5 L/d. Furthermore, the reduced organic load of the PABR effluent leads to the 
conclusion that can replace the benchmark approach for treating similar wastes, even though the biogas 
production is not suitable for energy recovery. For example, at the fourth experimental phase were the COD 
removal was the highest, the average daily bio-methane production was 30% lower from that of experimental 



phase 6. This approach would be efficient with liquid wastes of high organic load (10-6 g/L) if the PABR 
anaerobic system was used as a pretreatment method to aerobic oxidation tanks. 
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Abstract  

A microbial fuel cell (MFC) is a bioreactor that converts the chemical energy of the bonds of organic 
compounds to electrical energy, through the catalytic reactions of microorganisms. Under anaerobic 
conditions various substrates have been examined using MFC technology. This study examines the potential 
use of the liquid fraction of fermentable household waste (source-sorted food waste), which results from 
condensation of the vapors generated during drying, as a feed to the MFC. The main characteristics of this 
ǎǳōǎǘǊŀǘŜ ŀǊŜΥ мо Ǝ /h5κ[Σ ǇIҐоΦрΣ ŎƻƴŘǳŎǘƛǾƛǘȅҐнсн ˃{κŎƳΦ /ƻƴŘŜƴǎŀǘŜ ǿŀǎ ŦŜŘ ǘƻ ǘǿƻ ǎƛƴƎƭŜ-chamber air 
cathode MFCs, using mullite and GoreTex as cathodic electrodes, respectively. The oxygen reduction catalyst 
was MnO2 in both cases, while graphite granules were used as anodic electrodes. The units were operated in 
batch mode. Linear sweep voltammetry was carried out in order to conduct electrochemical characterization. 
The maximum ǇƻǿŜǊ ƻǳǘǇǳǘ ǿŀǎ лΦрн ˃²κƳ3 ŦƻǊ ǘƘŜ ƳǳƭƭƛǘŜ ŎŜƭƭ ŀƴŘ лΦну ˃²κƳ3 for the GoreTex cell, 
respectively. High COD removal efficiencies (>75%) were achieved for both cells.  

Introduction 

Efficient municipal waste management (MW) is a key factor that contributes to the overall waste management 
[European Commission Proposal 2015]. Although the amount of MW is not the same across EU countries 
[EUROSTAT 2017], food waste is the major fraction of MW [European Parliament News 2017], 53 % of which 
is produced in households [Stenmark 2016]. Food waste (FW) is rich in various substances with high energetic 
and nutritional content [Antonopoulou et al 2019], making an excellent supply for many biological processes 
[Ng et al 2020]. In the municipality of Halandri, in Attica, Greece, an innovative FW valorization approach has 
been developed and implemented at pilot-scale within the framework of the Horizon 2020 project 
WASTE4think [Antonopoulou et al 2019, Tremouli et al. 2019, Lytras et al. 2020]. The implemented waste 
management scheme included the source-separated collection of the household FW from 250 households. 
The collected FW was then led to a drying/shredding facility of the Municipality. The produced condensate is 
rich in organic carbon but poor in nitrogen which limits its biological treatment. An exploitation approach of 



condensate is its mixing with waste activated sludge as a feed for an anaerobic digester, in order to enhance 
the methane yield [Lytras et al. 2020]. 

Another approach of condensate exploitation is bioelectricity production using the microbial fuel cell (MFC) 
technology. This is an innovative approach to wastewater treatment with direct electricity output [Logan 
2009]. The MFC is comprised of an anode and a cathode compartment. In the anode the effluent is oxidized 
by bacteria, under anaerobic conditions. The produced electrons are collected through the anodic electrode 
and transferred to the cathodic electrode by an external resistance. The produced protons are transferred 
through the separator to the cathodic chamber. On the cathodic electrode an electron acceptor is reduced 
[Obileke et al. 2021]. Depending on the acceptor the presence of a catalyst may be needed; such is the case 
of O2. The separator between the anode and the cathode may increase the cost of the system, especially when 
membranes are used [Daud et al. 2015]. More feasible options than membranes have been researched and 
provided comparable results at lower costs, such as ceramic electrodes [Winfield et al. 2016] and GoreTex 
cloth [Tremouli et al. 2019].  

In this study, the performance of two single chamber MFCs with alternative cathode assemblies (ceramic and 
GoreTex) was studied. Two MFCs were acclimated and operated using raw condensate as the substrate.  

Materials and Methods 

Two similar Plexiglas single chamber MFCs were constructed, as described elsewhere [Tremouli et al. 2021]. 
Each cell had four tubes running through the anodic compartment. The same anode electrode setup was used 
for both cells, graphite granules (250 g) and an embedded graphite rod. Different cathode electrodes were 
used for the two cells. For the first cell, four cathodic electrodes with GoreTex cloth were assembled as 
specified in Tremouli et al. [2019]. For the second cell, the four cathodic mullite electrodes were internally 
coated with the oxygen reduction catalyst by mixing graphite paint (12 g), xylene (3 ml), ethanol (3 ml) and 3 
g MnO2Φ !ƴ ŜȄǘŜǊƴŀƭ ǊŜǎƛǎǘŀƴŎŜ ǎŜǘ ŀǘ млл ʍ ǿŀǎ ŎƻƴƴŜŎǘŜŘ ōŜǘǿŜŜƴ ǘƘŜ ŀƴƻŘŜ ŀƴŘ ǘƘŜ ŎŀǘƘƻŘŜ ŜƭŜŎǘǊƻŘŜǎΦ 

The cells were operated in batch mode, each filled up with 150 cm3 anodic liquid. The cells were placed in a 
temperature-controlled room, at 27 oC. During the acclimation period both cells were fed with synthetic 
wastewater consisting of phosphate buffer (3.67 g/L NaH2PO4 and 3.45 g/L Na2HPO4), potassium chloride (0.16 
g/L KCl), sodium bicarbonate (5 g/L NaHCO3), trace elements (1% v/v, described elsewhere [Skiadas & 
Lyberatos 1998]) and glucose (1.5 g COD/L) as the electron donor. During the first three acclimation cycles the 
cells were inoculated with anaerobic sludge (10 % v/v) obtained from the Likovrisi, Athens, Greece sewage 
treatment plant.   

Following the acclimation period, glucose was replaced with condensate. Condensate was produced by 
condensing the vapors that are generated during the drying and shredding of the pre-sorted fermentable 
fraction of household food waste collected door-to-door in the Municipality of Halandri, Greece [Ntaikou et al 
2018]. The characteristics of condensate were 13 g COD/L, pH=3.5, conŘǳŎǘƛǾƛǘȅҐнсн ˃{κŎƳ ŎƳ ŀƴŘ ŎƻƴǘŀƛƴŜŘ 
ǘƘŜ ŦƻƭƭƻǿƛƴƎ ǾƻƭŀǘƛƭŜ Ŧŀǘǘȅ ŀŎƛŘǎ ό±C!ǎύ ŀŎŜǘƛŎ мллу ҕ тнл ƳƎκ[Σ ǇǊƻǇƛƻƴƛŎ тр ҕ нр ƳƎκ[Σ ƛǎƻ-ōǳǘȅǊƛŎ пл ҕ ну 
ƳƎκ[Σ ōǳǘȅǊƛŎ мпп ҕ су ƳƎκ[ ŀƴŘ ƛǎƻ-ǾŀƭŜǊƛŎ мо ҕ п ƳƎκ[Φ ¢ƘŜ ±C!ǎ ŎƻƴŎŜƴǘǊŀǘƛƻƴǎ ŀǊŜ ǘƘŜ ŀǾŜǊŀƎŜ ǾŀƭǳŜǎ ƻŦ 
four different feed samples. In order to improve the low conductivity and pH, phosphate buffer was added in 
the raw condensate. After the mixing the improved pH and conductivity were 4.9 and 6.6 mS/cm, respectively. 
The condensate feeding presented fluctuations because it originated from gathered HFW, which varied each 
batch.  

The voltage of the cells was recorded every two minutes by a Keysight LXI Data Acquisition. Linear sweep 
voltammetry was conducted by a Potentiostat ς Galvanostat (PGSTAT128N ς AUTOLAB) with an Ag/AgCl 
reference electrode. The pH and conductivity were measured by digital instruments (WTW INOLAB PH720) 
and (WTW INOLAB) respectively.  Soluble COD was measured according to the standard methods [Standard 



Methods 2012]. For the quantification ƻŦ ±C!ǎΣ м Ƴƭ ƻŦ ǎŀƳǇƭŜ ŀŎƛŘƛŦƛŜŘ ǿƛǘƘ ол ˃[ ƻŦ нл҈ Iн{hп ǿŀǎ ŀƴŀƭȅȊŜŘ 
via a gas chromatograph (SHIMADZU GC-2010 plus) equipped with a flame ionization detector and a capillary 
ŎƻƭǳƳƴ ό!ƎƛƭŜƴǘ ǘŜŎƘƴƻƭƻƎƛŜǎΣ олƳ ȄлΦроƳƳ L5 Ȅм ˃Ƴ ŦƛƭƳΣ It-FFAP) using an auto sampler (SHIMADZU AOC-
20 s). 

Coulombic efficiency (CE) was calculated according to Eq.1. CE is the charge produced to the charge that is 
contained in the substrate and is calculated by [Logan 2009]: 

ὅὉ = 
Ͻ᷿

ϽϽϽ
 Eq. 1 

MO2 is the molar weight of Oxygen (=32), F is the Faraday constant (=96485 C/mol), b is number of electrons 
per O2 ƳƻƭŜ όҐпύΣ ± ƛǎ ǘƘŜ ŜŦŦŜŎǘƛǾŜ ǾƻƭǳƳŜ ƻŦ ǘƘŜ ŎŜƭƭ όҐмрл Ƴƭύ ŀƴŘ ɲ/h5 ƛǎ ǘƘŜ ŘƛŦŦŜǊŜƴŎŜ ōŜǘǿŜŜƴ ǘƘŜ ƛƴƛǘƛŀƭ 
and the final COD measurements for each batch cycle. 

Results 

GoreTex cell operation  

The duration of the acclimation period of the GoreTex cell was 660 h. For the acclimation to be considered 
complete, there had to be repeatable current peaks and high COD removal, implying development of the 
electrogenic active biofilm. The maximum current output for the acclimation cycles was 0.5 mA.  

Following the acclimation period, the glucose synthetic feed was replaced with raw condensate and the results 
are presented in Figure 1. Five cycles were carried out; the detailed results of each cycle are presented in Table 
1. The current output was comparable to the maximum current achieved during the acclimation, 
approximately 0.5 mA. However, the maximum current output was decreased over time, 1st cycle 0.62 mA and 
4th cycle 0.42 mA.  

Table 1 Measurements and calculations of the GoreTex cell operation. 

Cycle 
# 

CODin (g/L) 
Inlet 
pH 

Imax 

(mA) 
COD Removal (%) 

1st 9.9 5.9 0.62 95% 

2nd 9.2 4.0 0.52 96% 

3rd 12.7 4.9 0.44 94% 

4th 14.7 3.8 0.47 92% 

5th 13.8 4.3 0.42 77% 

The GoreTex cell achieved high COD removal (>92%) in all five cycles, but a decrease was observed in the last 
cycle (77%). The inlet characteristics in terms of COD and pH have possibly affected the performance of the 
cell, as the inlet COD is increased and the pH is lowered through the cycles. The maximum current output (0.62 
mA) was achieved in the first cycle, where both a low inlet COD (9.9 mg/L) and highest pH (5.9) were measured. 
Coulombic efficiencies calculated for the five cycles were very low, approximately 2%. 



 

Figure 1: Current output versus time during operation of the two cells (Black = GoreTex cell, Red = mullite cell). 
The cycles have been numbered with the respective colors. 

Mullite cell operation  

The duration of the acclimation of the mullite cell was 2010 h. The maximum current output for the acclimation 
cycles was 2.28 mA. Following the acclimation period, the synthetic glucose feed was replaced with raw 
condensate and the results are presented in Figure 1. Four cycles were carried out, the detailed results of each 
cycle being presented in Table 2. In particular, the maximum current output achieved was 2.02 mA, similar to 
the acclimation maximum current output (2.28 mA). However, the maximum current output decreased with 
time, as it can be seen in Figure 1 and Table 2 (current output for the 1st cycle 2.02 mA and 0.83 mA for the 4th 
cycle, respectively). As it can be seen from Figure 1, during the 3rd cycle the current output of the mullite cell 
(red line) presented fluctuations which are attributed to the electrical connection issues. 

The COD removal (>91%) was high for the mullite cell. The maximum current output is affected by the low pH 
and the high COD of the raw condensate while a drop is observed in its maximum value during time (2.02 mA 
1st cycle, 1.54 mA 2nd cycle, 1.26 mA 3rd cycle and 0.83 mA 4th cycle). Coulombic efficiencies calculated for the 
four cycles ranged between 4% - 9%. 

The VFAs were almost completely consumed by the microorganisms, leading to an increase in the pH in both 
cells at the end of every cycle (e.g. GoreTex cell 1st cycle inlet pH=5.9, outlet pH=7.13, mullite cell 4th cycle inlet 
pH 3.8, outlet pH=4.36). 

Table 2 Measurements and calculations of the mullite cell operation. 

Cycle # 
CODin 
(g/L) 

Inlet 
pH 

Imax 

(mA) 

COD Removal 
(%) 

1st 10.9 5.9 2.02 96% 

2nd 6.2 6.1 1.54 91% 

3rd 10.5 4.8 1.26 95% 

4th 12.5 3.8 0.83 94% 

Electrochemical Characterization 



 

Figure 2 Volumetric power density versus volumetric current density versus voltage as extracted by LSV 
experiment on the two cells. (Black = GoreTex cell, Red = mullite cell) 

Figure 2 presents the polarization curves of the two cells. The maximum power output was achieved by the 
GoreTex cell Pmax = 0.52 mW/m3, while the mullite cell achieved Pmax = 0.3 mW/m3. The OCVs (open circuit 
voltages) obtained were 0.228 V and 0.452 V for the GoreTex and mullite cell, respectively. The voltage versus 
volumetric current density curves indicate that, in both cells, ohmic resistances dominated, but were greater 
ŦƻǊ ǘƘŜ DƻǊŜ¢ŜȄ ŎŜƭƭΣ ŘǳŜ ǘƻ ŀ ƘƛƎƘŜǊ ǎƭƻǇŜ όсфу ʍ ƛƴǘŜǊƴŀƭ DƻǊŜ¢ŜȄ ŎŜƭƭ ǊŜǎƛǎǘŀƴŎŜΣ нмм ʍ ƛƴǘŜǊƴŀƭ ƳǳƭƭƛǘŜ ŎŜƭƭ 
resistance). Despite the facts that acclimation was not as fast and the OCV was not as high as for the GoreTex 
cell, the mulite cell overall performed better in terms of power output, waste treatment efficiency and 
coulombic efficiency.  

Conclusion 

The liquid fraction of dried fermentable household waste was treated using two single chamber MFCs using 
different cathode assemblies (GoreTex and mullite assembly, respectively). The mullite cell performed better 
than the GoreTex cell in terms of COD removal and power output. The results indicated that both cells had 
difficulty treating the raw condensate wastewater, because of its high COD 13 g/L and low pH 3.5 and 
conductivity 2.62 mS/cm.  
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2. Highlights 

¶ Revolutionize the field of waste and wastewater management 

¶ Develop an innovative approach in the field of municipal waste and wastewater management 

¶ Co-manage the liquid fraction of the Fermentable Municipal Solid Waste (condensate) and the 
Municipal Wastewater streams 

¶ PABR is a high-rate bioreactor, designed to operate under high organic loadings in low HRTs 

 

3. Purpose 

Historically, the management of municipal solid waste and that of wastewater have evolved in two 
independent streams, defined by the main phase in each case: solid and liquid, respectively. 

Currently, the benchmark approach to municipal wastewater management consists of sewer collection, 
treatment in a facility aiming at removal of suspended solids through primary sedimentation, biological 
oxidation of organic matter, biological nutrient (N and P) removal and disposal of the clarified effluent 
following disinfection by chlorination. The process generates a mixture of primary and excess secondary sludge 
which are typically mixed, stabilized by anaerobic digestion and dewatered before disposal [1,2]. 

Concerning solid waste management, for many years, the fermentable organic matter, mostly food waste, is 
collected usually mixed with recyclables in a commingled bin and then led to the landfill. Separate collection 
of food waste at the source allows for (a) high quality food waste that can be valorized and (b) cleaner 
recyclables [3]. 

The  purpose of present work is to develop an innovative approach in the field of municipal waste and 
wastewater management. More specifically, it focuses on the Fermentable Municipal Solid Waste (FMSW) and 
the Municipal Wastewater (MWW) streams. These two streams are currently being treated as separate waste 
streams; FMSW is collected as part of the mixed solid waste and landfilled and MWW is led to a wastewater 
treatment facility through the sewer. We hereby propose to explore an alternative that is much more 
meaningful and sustainable: To co-manage the liquid fraction of the Fermentable Municipal Solid Waste 
(condensate) and the Municipal Wastewater streams. We have recently developed a method to separate the 
solid and the liquid fraction of the Fermentable Municipal Solid Waste (drying and shredding process) and we 



aim to build upon this experience to develop an innovative and sustainable treatment framework that can 
revolutionize the field of waste and wastewater management [4]. 

4. Materials and methods 

The Periodic Anaerobic Buffled Reactor (PABR) is a novel bioreactor, designed to operate at high organic 
loading rates [5, 6, 7]. The PABR is a high-rate digester, suitable for municipal sewage treatment because of 
its low construction, operation and maintenance costs, low excess sludge production, and high capacity for 
biogas production at a small retention time.  

We evaluated the anaerobic digestion potential of a mixture of the condensate resulting from drying food 
waste and of municipal wastewater mixture using a PABR with an operating volume 77L. Biogas productivity, 
methane content, pH, alkalinity, sCOD, tCOD, TSS, VSS and VFAs were measured routinely. 

We developed also an activated sludge system (a sequencing batch reactor, SBR) for the treatment of the 
effluent of the PABR digester. The specific system was a 15L sequencing batch reactor. The SBR operation 
secure both COD and (N) removal. During the experiments soluble and total COD, total nitrogen (TN), alkalinity, 
TSS and VSS were measured regularly both in the feed and the effluent of the SBR. 

5. Results and discussion 

The PABR exhibited great stability in the effluent with a mean sCOD removal rate was 77%. The VSS remained 
below 520mg/L and TSS 610mg/L respectively. The mean biogas productivity was 51mg/L and the mean 
methane composition of the biogas was 66%. 

For the SBR, the pH levels in the effluent remained close to 8 (range between 7.7-8.3 - basic conditions), while 
the total alkalinity was around 1100mgCaCO3/L. SBR showed satisfactory behavior in removing organic loading 
below the environmental discharge limit of 120mg/L. The COD of the effluent remained approximately 100 
mg/L. Total nitrogen was 12.4mg/L while the TS and VSS were 0.25 mg/L and 0.2 mg/L respectively. 

6. Conclusions and perspectives 

Overall, the PABR proved to efficiently operate under HRTs as low as 4 days. Specifically, biogas production 
observed when the basic operational parameters were HRT 4 d and organic loading rate (OLR) 0.94gCOD/L*d. 
Biogas production was 51L/d. 

 

The work contributes to the development of a framework in which the liquid fraction (condensate) of the 
municipal fermentable household waste will be combined and co-managed with either the municipal 
wastewater in a common collection sewer system or will be transported and mixed with the excess sludge 
generated in conventional treatment plants, enhancing the generation of biogas. At the same time the dried 
and shredded food waste (food residue biomass) can be exploited in various ways as developed and described 
in [4]. 
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Abstract The objective of this work is the 
development of a new treatment scheme for 
Municipal Wastewater (MWW) along with the 
condensate from drying and shredding food waste. 
These two streams are currently collected and 
treated separately from each other. The new 
environmentally efficient treatment scheme 
proposed here is based on the chemical similarity of 
these two waste streams due to their mainly organic 
nature. Food waste was collected separately from 
households and led to a drier/shredder, where the 
liquid phase (condensate) is separated from the solid 
phase. The condensate is mixed with MWW and led 
to a high-rate anaerobic reactor, where both 
streams are treated. In this work a 77L high-rate 
digester, a Periodic Anaerobic Baffled Reactor 
(PABR) was utilized. The PABR effluent is fed in a 
Sequencing Batch Reactor (SBR) for further 
treatment until the final effluent meets the 
environmental criteria for safe disposal. The biogas 
generated from the PABR was 36.6 L/d with a 
methane content of 65%.  The SBR operated utilizing 
the Anaerobic ς Aerobic -Anoxic (AOA) process for 
simultaneous organic carbon and nutrient removal. 
The process generated an effluent suitable for 
disposal in terms of organic matter, nitrogen and 
phosphorus content. 

 

Keywords: Anaerobic Digestion; Food Waste; 
Municipal Wastewater; PABR; SBR 

5. Introduction 

Food Waste (FW) is defined as the food lost in the 
final stages of the Food Supply Chain (FSC), which 
includes the distribution and retail sector along with 
the final consumer [1]. Food loss is also observed 
along the previous food productions stages 
although, in the developed countries, households 
are responsible for most of the food waste [2]. 

Currently the majority of food waste in Greece is 
collected along with the rest of the non-recyclable 

municipal solid wastes and is led to landfills and 
incineration plants [3]. This management scheme 
results in biogas emissions from landfills impacting 
climate change and loss of potential energy or 
nutrient recovery from the fermentable fraction of 
municipal solid waste.  

On the other hand, Municipal Wastewater (MWW) 
is collected through the sewage system and led to 
Wastewater Treatment Plants (WWTP). The 
benchmark approach for MWW consists of primary 
sedimentation, biological oxidation utilizing aeration 
ŀƴŘ ǎŜŎƻƴŘŀǊȅ ǎŜŘƛƳŜƴǘŀǘƛƻƴΦ ¢Ƙƛǎ άŀŎǘƛǾŀǘŜŘ 
ǎƭǳŘƎŜέ ǇǊƻŎŜǎǎ ƛǎ ŎƘŀǊŀŎǘŜǊƛȊŜŘ ōȅ ƘƛƎƘ ŜƴŜǊƎȅ Ŏƻǎǘǎ 
especially for aeration and high biosolids generation 
[4]. Concerning food waste alternative treatments 
schemes along with food prevention policies are 
introduced in order to reduce its negative 
environmental impact. An alternative approach was 
ǇǊƻǇƻǎŜŘ ōȅ ǘƘŜ IƻǊƛȊƻƴ нлнл ά²ŀǎǘŜп¢Ƙƛƴƪέ 
research program where Food Residue Biomass 
(FORBI) resulting from drying and shredding food 
waste collected separately from households was 
valorized through Anaerobic Digestion for energy 
recovery among various alternatives [5]. 

Novel WW management schemes focus on 
processes with  lower energy consumption and 
higher nutrient recovery [6] AD is gaining ground, as 
it has low energy requirements but also the potential 
of energy recovery in the form of methane. While AD 
is currently used in WWTPs as a side process mostly 
for the treatment of the excess sludge generated by 
biological oxidation, there are recent projects 
examining the feasibility of AD as a direct MWW 
treatment process. A significant limitation is the low 
efficiency of anaerobic digestion for low organic 
strength wastewaters such as municipal 
wastewater.  High-rate AD systems may be used as 
the only viable option. A notable effort towards this 
goal is the installment of a full scale Expanded 
Granular Sludge Bed Reactor (EGSBR) in a WWTP in 
Ireland, during the operation of which, high BOD 
removal rate (85%) was accomplished [7].  It is, 
however, desirable to co-treat municipal 
wastewater with other higher strength wastewater 

mailto:kpapado@chemeng.ntua.gr


streams in the effort to increase the potential of 
methane recovery.  The liquid fraction of food waste 
is such a stream. 

While food waste is treated as solid waste, it 
contains high amounts of humidity (70-80%) [8] [9] 
[10] and MWW contains significant amount of 
biosolids [11]. This study is based on the approach 
ǇǊƻǇƻǎŜŘ ōȅ ǘƘŜ IƻǊƛȊƻƴ нлнл ά²!{¢9п¢Ƙƛƴƪέ 
research program  where the drying process divided 
food waste in two streams: a) the solid part called 
FORBI and b) the liquid part containing the majority 
of the moisture content of food waste and other 
volatile compounds such as Volatile Fatty Acids 
(VFAs) and Ethanol [12].  

It is then a reasonable option to consider the co-
treatment of MWW with the condensate generated 
upon drying food waste in a high-rate anaerobic 
digester.  The high-rate anaerobic system utilized in 
this study is the Periodic Anaerobic Baffled Reactor 
(PABR), while the digestate was treated in a 
Sequencing Batch Reactor (SBR) in order to attain 
effluent characteristics that are suitable for 
discharge to receiving water bodies.  

 

6. Materials and Methods 

A synthetic wastewater for testing the idea of co-
digestion  was formulated based on the literature 
and had the following composition: 0.7g/L glucose, 
0.019 g/L NH4Cl, 0.08 g/L CH3COONa, 0.044 g/L 
KH2PO4, 0.0275 g/L MgSO4*7H2O, 0.0025 g/L CaCl2, 
0.004 g/L KCl, 0.125 g/L NaHCO3, 1.875 mg/L 
FeCL3*6H2O, 0.1875 mg/L H3BO3, 0.225 mg/L KI, 0.15 
mg/L MnSO4, 0.0275 mg/L ZnSO4*7H2O, 0.0375 
mg/L CuSO4*5H2O and 12.5 mg/L EDTA (Liu et al. 
2013). The FW condensate resulting from drying and 

shredding FW for the restaurant of the National 
technical University of Athens. FW was collected 
separately daily and led to a dryer where the liquid 
part (condensate) was separated from the solid 
residue. The condensate was then mixed with the 
synthetic municipal wastewater at a ratio of 10% 
/90% respectively and fed to a PABR. The PABR is a 
high-rate anaerobic system developed by Skiadas 
and Lyberatos (Skiadas, I. V., Lyberatos 1998).  It 
consists of two concentric cylinders as shown in 
Figure 1, with the inner one operating as heat 
exchanger. The space between the two cylinders is 
divided into four compartments, each of which is 
divided in two sections, one down- flow and one up-
flow, thus resembling a simple Anaerobic Baffled 
Reactor (ABR), only arranged in a circular structure. 
The innovative approach of this bioreactor is its 
ability to periodically change the inflow and outflow 
compartments. The Switching Period T (the period 
for one complete switching of compartment roles), 
along with the Hydraulic Retention Time (HRT) 
determine the flow patterns of the reactor. In this 
work, a 77L PABR was utilized and operated under 
ABR mode (compartment switching was not 
imposed.). From previous experiments the PABR 
showed higher efficiency both in terms of COD 
removal and biogas production operating without 
Switching Period imposed for this type of 
waste(Zarkaliou et al. 2022). 

In this work a 15 L SBR was used. to treat the PABR 
effluent so as to reach the required effluent quality 
for discharge. The SBR had an external jacket for 
temperature control via water circulation. Every 
operation cycle of the SBR was divided in four 
phases. The first phase was the feeding phase 
followed by the reaction period that consisted of 
distinct Anaerobic, Aerobic and Anoxic conditions. 
The final phases of the operation cycle were the 
sedimentation phase and the discharge.  



 

 

Figure 3 PABR experimental set-up 

 

 

The PABR operated under 1 day HRT with no 
switching period imposed. The reactor was fed 
with 77L of feedstock daily. The Feedstock was a 
mixture of 90% synthetic Municipal Wastewater 
and 10% Fermentable Municipal Solid Waste 
condensate. The temperature was kept constant at 
35 oC.  The effluent of the PABR was then mixed 
again with condensate in a ration of 85 % PABR 
effluent and 15% condensate and led to the SBR 
reactor. The reactor was fed at the 2/3 of its active 
volume in every cycle. The 1/3 of the volume was 
constantly occupied by the activated sludge. The 
operational parameters of the Anaerobic Digestion 
process are presented in Table 1 and those of the 
Activated Sludge Process in Table 2. 

 

3. Results and Discussion 

Concerning the AD process, the mixture of FW 
condensate and synthetic MWW constituted a 
suitable feedstock for the PABR. Especially in terms 
of biogas the average daily production reached the 
value of 36.6 L/d with a methane content of 65 %. 
In terms of COD removal, the process reached a 
rate of 92% which is similar to that of biological 
oxidation process but without consuming 
equivalent energy. Figure 2 shows the daily biogas 
and biomethane production. While the digestate 
COD was low other nutrients had concentrations 
higher than the environmental limit. Especially 
Nitrogen (TN) was close to 50 mg/L, while Total 
Phosphorus (TP) between 4-5 mg/L. The digestate 
was led to the SBR in order to achieve a total TN 
below 12 mg/L and TP below 2 mg/l. After the 
Anaerobic-Aerobic- Anoxic (AOA) process in the 
SBR the final effluent had an average TN and TP 

value of 6 mg/L and 1.9 mg/L respectively. The 
daily measurements of the TN concentration in the 
SBR feed and effluent are shown in Figure 3. 

It can be concluded that while the Anaerobic co ς 
digestion of the FW condensate along with 
municipal wastewater can reduce their organic 
content at a sufficient rate, it is not effective 
regarding other nutrients such as nitrogen and 
phosphorus. However, high-rate AD could serve as 
a first step before the activated sludge process in 
²²¢tΩǎ ŀƴŘ ǊŜŘǳŎŜ ǘƘŜ ŜƴŜǊƎȅ ŎƻƴǎǳƳǇǘƛƻƴ ƻŦ 
aeration needed for the oxidation of the organic 
matter while also producing methane. 



 
Table 3 Operational parameters of the Anaerobic Digestion process. 

HRT  1 d 

Switching Period ɇ  - 

Organic Loading Rate OLR  1,035 ( gCOD/L/d) 

Feedstock Temperature  4 (ɞC) 

Reactor Temperature  35 (ɞC) 

 

 

Table 4. Operational parameters of biological oxidation process 

HRT  0.69 d 

Cycle Duration  11 hrs 

Feedstock Fed per Cycle  10 L 

Feedstock Temperature  4 (ɞC) 

Reactor Temperature  25 (ɞC) 

  

 

    
 

                                         Figure 4. Daily production of biogas and bio-methane in the PABR 
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                                                     Figure 5. TN concentration in SBR feedstock and effluent. 

4. Conclusion 

This work proposed a novel management scheme where two municipal waste streams are co-treated by 
utilizing their similar chemical nature (organic) in comparison with the physical phase approach which has 
been traditionally used for collection and disposal of solid and liquid wastes, respectively.  

The proposed novel management scheme includes the co-digestion of FW condensate with MWW in a 
high-rate anaerobic system with potential energy recovery and treatment of the digestate by 
biological oxidation in order to reach the environmental safety disposal criteria. 
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˃ʷˋʹ ˉʰˊʰʴ˖ʴʽˁˈˍʹˍʰ ʲʽˇʰʶˊʾˇˎ орΦу [κŘ ˁʰʽ ˉˇˋˇˋˍˈ ˃ʶʻʰ˄ʾˇˎ сп҈Φ ɮˉˇʵʶʾ˔ʻʹˁʶ ˈˍʽ ˇ t!.w ˃ˉˇˊʶʾ 
˄ʰ ˂ʶʽˍˇˎˊʴʺˋʶʽ ʰˉˇˍʶ˂ʶˋ˃ʰˍʽˁʱ ˃ʶ ˎʵˊʰˎ˂ʽˁˈ ˔ˊˈ˄ˇ ˉʰˊʰ˃ˇ˄ʺˌ όIw¢ύ ʷ˖ˌ ˁʰʽ м ʹ˃ʷˊʰΦ  
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1. ɳɹʅɮɱʍɱɶ 

ʅʺ˃ʶˊʰ ʹ ˁˏˊʽʰ ˉˊˇˋʷʴʴʽˋʹ ˍʹˌ ʵʽʰ˔ʶʾˊʽˋʹˌ ˍ˖˄ !̀ ˍʽˁ˗˄ ʇʴˊ˗˄ !̄ ˇʲ˂ʺˍ˖˄ ʰ˒ˇˊʱ ˍʹ ˋˎ˂˂ˇʴʺ ˍˇˎˌ 
˃ʷˋ˖ ʰˉˇ˔ʶˍʶˎˍʽˁˇˏ ʵʽˁˍˏˇˎ ˁʰʽ ˍʹ˄ ʶˉʶ˅ʶˊʴʰˋʾʰ ˍˇˎˌ ˋʶ ˃ˇ˄ʱʵʶˌ ʲʽˇ˂ˇʴʽˁˇˏ ˁʰʻʰˊʽˋ˃ˇˏ ʴʽʰ ˍʹ˄ 
ʰ˒ʰʾˊʶˋʹ ˍ˖˄ ʰʽ˖ˊˇˏ˃ʶ˄˖˄ ˋˍʶˊʶ˗˄ ˁʰʽ ˍ˖˄ ʻˊʶˉˍʽˁ˗˄ ˋˎˋˍʰˍʽˁ˗˄ ˍˇˎˌ όʱʸ˖ˍˇΣ ˒˗ˋ˒ˇˊˇύ ˃ʷˋ˖ 
ʵʽʶˊʴʰˋʽ˗˄ ˁʰʻʾʸʹˋʹˌΣ ʲʽˇ˂ˇʴʽˁʺˌ ˇ˅ʶʾʵ˖ˋʹˌ ˍˇˎ ˇˊʴʰ˄ʽˁˇˏ ˒ˇˊˍʾˇˎ ˁʰʽ ʰˉˇ˂ˏ˃ʰ˄ˋʹˌ ˃ʷˋ˖ 
˔˂˖ˊʾ˖ˋʹˌΦ ʅˍʹ ˋˎ˄ʷ˔ʶʽʰ ʷ˄ʰ ˃ʶʾʴ˃ʰ ˉˊ˖ˍˇʴʶ˄ˇˏˌ ˁʰʽ ʵʶˎˍʶˊˇʴʶ˄ˇˏˌ ʽ˂ˏˇˌ ˉˇˎ ˉʰˊʱʴʶˍʰʽ ʰˉˈ ˍʹ˄ 
ˉʰˊʰˉʱ˄˖ ʵʽʶˊʴʰˋʾʰ ˇʵʹʴʶʾˍʰʽ ʴʽʰ ʰ˄ʰʶˊˈʲʽʰ ˔˗˄ʶˎˋʹ ˉˊʽ˄ ˍʹ˄ ˍʶ˂ʽˁʺ ˍˇˎ ʵʽʱʻʶˋʹΦ ʆˇ ˁˏˊʽˇ 
˂ʶʽˍˇˎˊʴʽˁˈ ˁˈˋˍˇˌ ˉˇˎ ˒ʷˊʶʽ ˍˇ ˉʰˊʰˉʱ˄˖ ˋˏˋˍʹ˃ʰ ʶˉʶ˅ʶˊʴʰˋʾʰˌ ʲʰˋʾʸʶˍʰʽΣ ˋˍˇ˄ ʰʶˊʽˋ˃ˈ ˍʹˌ ʽ˂ˏˇˌ 
ˉˇˎ ʰˉʰʽˍʶʾˍʰʽ ʴʽʰ ˍʹ˄ ˇ˅ʶʾʵ˖ˋʹ ˍˇˎ ˇˊʴʰ˄ʽˁˇˏ ̱ʹˌ ˒ˇˊˍʾˇˎ ώмϐΦ ɱʽʰ ˍʹ˄ ʶ˅ˇ˃ʱ˂ˎ˄ˋʹ ˍˇˎ ˋˎʴˁʶˁˊʽ˃ʷ˄ˇˎ 
ˁˈˋˍˇˎˌ ʷ˔ʶʽ ˉˊˇˍʰʻʶʾΣ ˋˎ˄ˇ˂ʽˁʺ ʺ ˃ʶˊʽˁʺ ˋˍʰʻʶˊˇˉˇʾʹˋʹ ˍ˖˄ ʰˉˇʲ˂ʺˍ˖˄ ˃ʷˋ˖ ʰ˄ʰʶˊˈʲʽʰˌ ˔˗˄ʶˎˋʹˌ 
˃ʶ ˋˁˇˉˈ ˍʹ˄ ˉʰˊʰʴ˖ʴʺ ʶ˄ʷˊʴʶʽʰˌ ώнϐΦ ʍˋˍˈˋˇΣ ʹ ˋˎʴˁʶˁˊʽ˃ʷ˄ʹ ˉˊˈˍʰˋʹ ʷ˔ʶʽ ʰˉˇˍʶ˂ʶˋ˃ʰˍʽˁʺ ʶ˒ʰˊ˃ˇʴʺ 
˃ˈ˄ˇ ʴʽʰ ˉʶˊʽˇ˔ʷˌ ˃ʶ ʻʶˊ˃ˈ ˁ˂ʾ˃ʰΣ ˂ˈʴ˖ ˍʹˌ ˔ʰ˃ʹ˂ʺˌ ˋˎʴˁʷ˄ˍˊ˖ˋʹˌ ˋʶ ˇˊʴʰ˄ʽˁˈ ˒ˇˊˍʾˇ ˉˇˎ ˉʶˊʽʷ˔ʶˍʰʽ 
ˋˍˇ˄ ˋˎʴˁʶˁˊʽ˃ʷ˄ˇ ˍˏˉˇ ʰˉˇʲ˂ʺˍ˖˄ Φ 

ʂˋˇ˄ ʰ˒ˇˊʱ ˋˍʹ ʵʽʰ˔ʶʾˊʽˋʹ ˍ˖˄ ʲʽˇʰˉˇʽˁˇʵˇ˃ʺˋʽ˃˖˄ ˋˍʶˊʶ˗˄ ʰˉˇʲ˂ʺˍ˖˄ ˁʰʽ ʽʵʽʰʾˍʶˊʰ ˍ˖˄ ˇʽˁʽʰˁ˗˄ 
ˍˊˇ˒ʽˁ˗˄ ʰˉˇʲ˂ʺˍ˖˄Σ ʷ˔ʶʽ ʵʽʰˉʽˋˍ˖ʻʶʾ ˈˍʽ ˉʶˊʽʷ˔ˇˎ˄ ʰˊˁʶˍʱ ˎ˕ʹ˂ˈ ˎʴˊˈ ˉʶˊʽʶ˔ˈ˃ʶ˄ˇΣ όˍˎˉʽˁʱ тлπул҈ύ 
ʵʽʰ˔ʶʾˊʽˋʺ ˍˇˎˌ ʰˉˈ ˍʰ ˎˉˈ˂ˇʽˉʰ ˋˍʶˊʶʱ ʰˉˈʲ˂ʹˍʰ ˃ʶ ʰˉˇˍʷ˂ʶˋ˃ʰ ˄ʰ ˋˎ˃ʲʱ˂ˇˎ˄ ʰˊ˄ʹˍʽˁʱ ˍˈˋˇ ˋˍʹ˄ 
ʰˉ˗˂ʶʽʰ ˋʹ˃ʰ˄ˍʽˁˇˏ ˉˇˋˇˋˍˇˏ ˁʰʻʰˊ˗˄ ʰ˄ʰˁˎˁ˂˗ˋʽ˃˖˄ ˎ˂ʽˁ˗˄ ˈˋˇ ˁʰʽ ˋˍʹ˄ ʰˉʶ˂ʶˎʻʷˊ˖ˋʹ ˃ʶʴʱ˂˖˄ 
ˉˇˋˇˍʺˍ˖˄ ʲʽˇʰʶˊʾˇˎ ˃ʷˋ˖ ˍ˖˄ ʋʇʆɮΣ ˋˎ˃ʲʱ˂˂ˇ˄ˍʰˌ ˋʹ˃ʰ˄ˍʽˁʱ ˋˍˇ ˒ʰʽ˄ˈ˃ʶ˄ˇ ˍˇˎ ʻʶˊ˃ˇˁʹˉʾˇˎΦ 
ɼˇʽ˄ˇˍʽˁʷˌ ʁʵʹʴʾʶˌ όɳˎˊ˖ˉʰʿˁʺ ɴ˄˖ˋʹύΣ ˉʽʷʸˇˎ˄ ˉˊˇˌ ˍʹ˄ ˇˊʻˇ˂ˇʴʽˁʺ ʵʽʰ˔ʶʾˊʽˋʹ ˍˇˎ ˋˎʴˁʶˁˊʽ˃ʷ˄ˇˎ 
ˍˏˉˇˎ ʰˉˇʲ˂ʺˍ˖˄ ˉˇˎ ʰ˒ˇˊˇˏ˄ ʰˊ˔ʽˁʱ ˍʹ ˔˖ˊʽˋˍʺ ˋˎ˂˂ˇʴʺ ˍˇˎˌ ˋˍʹ ˉʹʴʺ  ˁʰʽ ˍʹ˄ʵʽʰ˒ˇˊˇˉˇʾʹˋʺ ˍˇˎˌ 
ʰˉˈ ˍˇ ˊʶˏ˃ʰ ˉˇˎ ˇʵʹʴʶʾˍʰʽ ˋˍˇˎˌ ʋʇʆɮ  ˃ʶ ˋˁˇˉˈ ˍʹ˄ ʶ˄ʶˊʴʶʽʰˁʺ ˍˇˎˌ ʰ˅ʽˇˉˇʾʹˋʹ ˃ʷˋ˖ ˃ʶʻˈʵ˖˄ ˈˉ˖ˌ 
ʰˎˍʺ ˍʹˌ ʰ˄ʰʶˊˈʲʽʰˌ ˔˗˄ʶˎˋʹˌ ώпΣ рϐΦ  

ɴ˄ʰ ˉˊ˗ˍˇ ˋˍʱʵʽˇ ʴʽʰ ˍʹ˄ ˇˊʻˇ˂ˇʴʽˁʺ ʰ˅ʽˇˉˇʾʹˋʹ ˍ˖˄ ʲʽˇʰˉˇʽˁˇʵˇ˃ʺˋʽ˃˖˄ ˋˍʶˊʶ˗˄ ʰˉˇʲ˂ʺˍ˖˄ 
ʶˉʽˍʶˏ˔ʻʹˁʶ ˋˍˇ ˉ˂ʰʾˋʽˇ ˍˇˎ ʃˊˇʴˊʱ˃˃ʰˍˇˌ IƻǊƛȊƻƴнлнл ό ²ŀǎǘŜп¢Ƙƛƴƪύ  ̍̄ ˇˎ ˍˇ ɳɾʃΣ ˋʶ ˋˎ˄ʶˊʴʰˋʾʰ 
˃ʶ ˍˇ˄ ɲʺ˃ˇ ʋʰ˂ʰ˄ʵˊʾˇˎ ʶ˒ʱˊ˃ˇˋʰ˄ ʷ˄ʰ ˉʽ˂ˇˍʽˁˈ ˉˊˈʴˊʰ˃˃ʰ ˔˖ˊʽˋˍʺˌ ˋˎ˂˂ˇʴʺˌ ˍ˖˄ ˇʽˁʽʰˁ˗˄ 
ˍˊˇ˒ʽˁ˗˄ ʰˉˇʲ˂ʺˍ˖˄Σ ˉˊˇπʁ ˉʶ˅ʶˊʴʰˋʾʰˌ ˁʰʽ ʶ˄ʶˊʴʶʽʰˁʺˌ ʰ˅ʽˇˉˇʾʹˋʹˌ ʰˎˍ˗˄ ˃ʷˋ˖ ʰ˄ʰʶˊˈʲʽʰˌ 
˔˗˄ʶˎˋʹˌΦ ɶ ˉˊˇπʁ ˉʶ˅ʶˊʴʰˋʾʰ ʰ˒ˇˊʱ ˍˇ˄ ˍʶ˃ʰ˔ʽˋ˃ˈ ˁʰʽ ˍʹ˄ ˅ʺˊʰ˄ˋʹ ˍʹˌ ˃ʱʸʰˌ ˍ˖˄ ʰˉˇʲ˂ʺˍ˖˄ 
˔˖ˊʾʸˇ˄ˍʱˌ ˍˇ ̀ʶ ʵˏˇ ˊʶˏ˃ʰˍʰΣ ˍˇ ˋˍʶˊʶˈ όCƻƻŘ wŜǎƛŘǳŜ .ƛƻƳŀǎǎΣ Chw.Lύ ˉˇˎ ˇʵʹʴʶʾˍʰʽ ʴʽʰ ʰ˄ʰʶˊˈʲʽʰ 
˔˗˄ʶˎˋʹ ˁʰʽ ˍˇ ʰˊˁʶˍʱ ˉ˂ˇˏˋʽˇ ˋʶ ˇˊʴʰ˄ʽˁˈ ˒ˇˊˍʾˇ ˎʴˊˈ ˋˎ˃ˉˏˁ˄˖˃ʰ όŎƻƴŘŜƴǎŀǘŜύΦ ɶ ˉʰˊˇˏˋʰ 
ʶˊʴʰˋʾʰ  ˋˍˇ˔ʶˏʶʽ ˋˍʹ˄ ˋˎ˄ʶˉʶ˅ʶˊʴʰˋʾʰ ˍˇˎ ˎʴˊˇˏ ʰˎˍˇˏ ˁ˂ʱˋ˃ʰˍˇˌ ˁʰʽ ʶ˄ʶˊʴʶʽʰˁʺ ʰ˅ʽˇˉˇʾʹˋʺ ˍˇˎ 
˃ʷˋ˖ ʰ˄ʰʶˊˈʲʽʰˌ ˔˗˄ʶˎˋʹˌ ˃ʰʸʾ ˃ʶ ˋˎ˄ʻʶˍʽˁˈ ʰˉˈʲ˂ʹˍˇ ˉˇˎ ˉˊˇˋˇ˃ˇʽʱʸʶʽ ˍʰ ˔ʹ˃ʽˁʱ ˔ʰˊʰˁˍʹˊʽˋˍʽˁʱ 
ˍ˖˄ ˁˇʽ˄˗˄ ˎʴˊ˗˄ ʰˋˍʽˁ˗˄ ˂ˎ˃ʱˍ˖˄Φ    

ɱʽʰ ˍʹ˄ ˃ʶ˂ʷˍʹ ˍʹˌ ˉʰˊʰˉʱ˄˖ ʵʽʶˊʴʰˋʾʰˌ ʶˉʽ˂ʷ˔ʻʹˁʶ ʷ˄ʰˌ ʃʶˊʽˇʵʽˁˈˌ ɮ˄ʰʶˊˈʲʽˇˌ ɮ˄ˍʽʵˊʰˋˍʺˊʰˌ ˃ʶ 
ɮ˄ʰˁ˂ʰˋˍʺˊʶˌ ˍˏˉˇˎ t!.w ό tŜǊƛƻŘƛŎ !ƴŀŜǊƻōƛŎ .ŀŦŦƭŜŘ wŜŀŎǘƻǊύ ώсΣ тΣ уΣ фϐΦ ʁ t!.w ʶʾ˄ʰʽ ʷ˄ʰˌ ˁʰʽ˄ˇˍˈ˃ˇˌ 
ˍʰ˔ˏˊˊˎʻ˃ˇˌ ʰ˄ˍʽʵˊʰˋˍʺˊʰˌ ˁʰʽ ˃ʽʰ ʶ˂ˁˎˋˍʽˁʺ ʵʽʰʵʽˁʰˋʾʰ ʴʽʰ ˍʰ ʰˋˍʽˁʱ ˂ˏ˃ʰˍʰ ˂ˈʴ˖ ˍˇˎ ˔ʰ˃ʹ˂ˇˏ 
ˁˈˋˍˇˎˌ ˁʰˍʰˋˁʶˎʺˌΣ ˂ʶʽˍˇˎˊʴʾʰˌ ˁʰʽ ˋˎ˄ˍʺˊʹˋʹˌΣ ˍʹˌ ˔ʰ˃ʹ˂ʺˌ ˉʰˊʰʴ˖ʴʺˌ ʽ˂ˏˇˌ ˁʰʽ ˍʹˌ ˎ˕ʹ˂ʺˌ 
ˉʰˊʰʴ˖ʴʺˌ ʲʽˇʰʶˊʾˇˎ ˋʶ ˃ʽˁˊˈ ˔ˊˈ˄ˇ ˉʰˊʰ˃ˇ˄ʺˌ ˍˈˋˇ ˋʶ ˎ˕ʹ˂ʷˌ ˈˋˇ ˁʰʽ ˋʶ ˔ʰ˃ʹ˂ʷˌ ˇˊʴʰ˄ʽˁʷˌ 
˒ˇˊˍʾˋʶʽˌΦ ʁ t!.w hˉˇˍʶ˂ʶʾˍʰʽ ʰˉˈ ʵˏˇ ˇ˃ˈˁʶ˄ˍˊˇˎˌ ˁˎ˂ʾ˄ʵˊˇˎˌΣ ˈˉ˖ˌ ˒ʰʾ˄ʶˍʰʽ ˋˍʹ˄ όɳʽˁˈ˄ʰ мύΣ ˃ʶ ˍˇ˄ 
ʶˋ˖ˍʶˊʽˁˈ ˄ʰ ˂ʶʽˍˇˎˊʴʶʾ ˖ˌ ʶ˄ʰ˂˂ʱˁˍʹˌ ʻʶˊ˃ˈˍʹˍʰˌΣ ˗ˋˍʶ ˄ʰ ʵʽʰˍʹˊʶʾˍʰʽ ˍˇ ˋˏˋˍʹ˃ʰ ˋʶ ˋˍʰʻʶˊʺ 
ʻʶˊ˃ˇˁˊʰˋʾʰΦ ʁ ˔˗ˊˇˌ ʰ˄ʱ˃ʶˋʰ ˋˍˇ˄ ʶˋ˖ˍʶˊʽˁˈ ˁʰʽ ˍˇ˄ ʶ˅˖ˍʶˊʽˁˈ ˁˏ˂ʽ˄ʵˊˇ ʰˉˇˍʶ˂ʶʾˍʰʽ ʰˉˈ ˍʷˋˋʶˊʰΣ 
ʾʵʽˇˎ ˈʴˁˇˎ ʵʽʰ˃ʶˊʾˋ˃ʰˍʰΣ ˁʱʻʶ ʷ˄ʰ ʰˉˈ ˍʰ ˇˉˇʾʰ ˔˖ˊʾʸʶˍʰʽ ˋʶ ʷ˄ʰ ʰ˄ˇʵʽˁˈ ˁʰʽ ʷ˄ʰ ˁʰʻˇʵʽˁˈ ˍˇ˃ʷʰΦ 
ɼʱʻʶ ʵʽʰ˃ʷˊʽˋ˃ʰ ˉˊˇˋˇ˃ˇʽʱʸʶʽ ˎʵˊʰˎ˂ʽˁʱΣ ʷ˄ʰ˄ ʰ˄ʰʶˊˈʲʽˇ ʰ˄ˍʽʵˊʰˋˍʺˊʰ ˃ʶ ʰ˄ʰˁ˂ʰˋˍʺˊʶˌ !.w 



ˉʰˊʰ˃ʷˍˊˇˎ ˂ʶʽˍˇˎˊʴʾʰˌ ʹ ˇˉˇʾʰ ˇ˄ˇ˃ʱʸʶˍʰʽ ʃʶˊʾˇʵˇˌ ɳ˄ʰ˂˂ʰʴʺˌ ό{ǿƛǘŎƘƛƴƎ tŜǊƛƻŘ ύ ¢Φ ɶ ˋˎʴˁʶˁˊʽ˃ʷ˄ʹ 
ˉʰˊʱ˃ʶˍˊˇˌ ˁʰʻˇˊʾʸʶʽ ˍˇ ˔ˊˇ˄ʽˁˈ ʵʽʱˋˍʹ˃ʰ ʴʽʰ ˍˇ ˇˉˇʾˇ ˁʱʻʶ ʵʽʰ˃ʷˊʽˋ˃ʰ ˍˇˎ ʰ˄ˍʽʵˊʰˋˍʺˊʰ ˂ʶʽˍˇˎˊʴʶʾ 
˖ˌ ʵʽʰ˃ʷˊʽˋ˃ʰ ɳʽˋˊˇʺˌΦ ɶ ˉʶˊʾˇʵˇˌ ʶ˄ʰ˂˂ʰʴʺˌ ʆ ˃ʰʸʾ ˃ʶ ˍˇ˄ ˎʵˊʰˎ˂ʽˁˈ ˔ˊˈ˄ˇ ˉʰˊʰ˃ˇ˄ʺˌΣ IȅŘǊŀǳƭƛŎ 
˔ʰ˃ʹ˂ʷˌ ˍʽ˃ʷˌ ˍʹˌ ˉʰˊʰ˃ʷˍˊˇˎ ʆ ˇ t!.w ̄ ˊˇˋʶʴʴʾʸʶʽ ˍʹ˄ ˋˎ˃ˉʶˊʽ˒ˇˊʱ ʶ˄ˈˌ ʰ˄ˍʽʵˊʰˋˍʺˊʰ ¦ǇŦƭƻǿ 
˃ʶˍʰ˅ˏ ˍ˖˄ ʵʽʰ˃ʶˊʽˋ˃ʱˍ˖˄Σ ˇ ʰ˄ˍʽʵˊʰˋˍʺˊʰˌ ˉˊˇˋˇ˃ˇʽʱʸʶʽ ˍʹ ˋˎ˃ˉʶˊʽ˒ˇˊʱ ˍʶˋˋʱˊ˖˄ ʰ˄ˍʽʵˊʰˋˍʺˊ˖˄ 
!.w ̀ˎ˄ʵʶʵʶ˃ʷ˄˖˄ ˋʶ ˋʶʽˊʱΦ ʅˍʹ˄ ˋˎʴˁʶˁˊʽ˃ʷ˄ʹ ʶˊʴʰˋʾʰ ʶ˅ʶˍʱʸʶˍʰʽ ʹ ʰˉˇˍʶ˂ʶˋ˃ʰˍʽˁˈˍʹˍʰ ˍˇˎ t!.wΣ ˖ˌ 
ˉˊˇˌ ˍʹ˄ ʰ˄ʰʶˊˈʲʽʰ ˋˎʴ˔˗˄ʶˎˋʹ ˃ʾʴ˃ʰˍˇˌ !̀ ˍʽˁ˗˄ ʇʴˊ˗˄ !̄ ˇʲ˂ʺˍ˖˄Σ ƳǳƴƛŎƛǇŀƭ ǿŀǎǘŜǿŀǘŜǊ όa²²ύ 
ˁʰʽ ʅˎ˃ˉˎˁ˄˗˃ʰˍˇˌ ʆˊˇ˒ʽˁ˗˄ ɮˉˇʲ˂ʺˍ˖˄ όŎƻƴŘŜƴǎŀǘŜύΦ 

 

2. ɾɳɸʁɲʁɽʁɱɹɮ 

ʋˊʹˋʽ˃ˇˉˇʽʺʻʹˁʶ ˋˎ˄ʻʶˍʽˁˈ ˃ʾʴ˃ʰ ˉˇˎ ˉˊˇˋˇ˃ˇʽʱʸʶʽ ˍʰ ˒ˎˋʽˁˇ˔ʹ˃ʽˁʱ ˔ʰˊʰˁˍʹˊʽˋˍʽˁʱ ˍ˖˄ a²²Φ ɶ 
ˋˎ˄ˍʰʴʺ ʴʽʰ ˍˇ ˋˎʴˁʶˁˊʽ˃ʷ˄ˇ ʵʽʱ˂ˎ˃ʰ ʰ˄ˍ˂ʺʻʹˁʶ ʲʽʲ˂ʽˇʴˊʰ˒ʽˁʱ ˁʰʽ ʶʾ˄ʰʽ ʹ ʶ˅ʺˌΥ  лΣтƎκ[  /сIмнhс Σ лΦмнп 
Ǝ
[
b
I
/
ƭ
Ǝ
[
/
I
/
h
h
b
ŀ
Ǝ
[
Y
I
t
h
Ǝ
[
a
Ǝ
{
h
I
h
Ǝ
[
/
ŀ
/
ƭ
Ǝ
[

ʁʽ ˉʰˊʱ˃ʶˍˊˇʽ ˂ʶʽˍˇˎˊʴʾʰˌ ʴʽʰ ˁʱʻʶ ˉʶʽˊʰ˃ʰˍʽˁʺ ˒ʱˋʹ ˉʰˊˇˎˋʽʱʸˇ˄ˍʰʽ ˋˍˇ˄ ʃʾ˄ʰˁʰ мΦ ʁ ʰ˄ˍʽʵˊʰˋˍʺˊʰˌ 
˂ʶʽˍˇˏˊʴʹˋʶ ˔˖ˊʾˌ ˉʶˊʽˇʵʽˁʺ ʶ˄ʰ˂˂ʰʴʺ ό˖ˌ !.wύΦ ɼʰʻΩ ̍˂ ʹ ˍʹ˄ ʵʽʱˊˁʶʽʰ ˍ˖˄ ˉʶʽˊʰ˃ʱˍ˖˄ ʵʽʶ˅ʱʴˇ˄ˍʰ˄ 
ˁʰʻʹ˃ʶˊʽ˄ʷˌ ʰ˄ʰ˂ˏˋʶʽˌ  ʵʽʰ˂ˎˍˇˏ ˔ʹ˃ʽˁ˗ˌ ʰˉʰʽˍˇˏ˃ʶ˄ˇˎ ˇ˅ˎʴˈ˄ˇˎ όǎ/h5ύΦ I ̄ʰˊʰʴ˖ʴʺ ʲʽˇʰʶˊʾˇˎ ˁʰʽ 
ɾʶʻʰ˄ʾˇˎ ˉʰˊʰˁˇ˂ˇˎʻˇˏ˄ˍʰ˄ ʹ˃ʶˊʹˋʾ˖ˌΦ ʂ˂ʶˌ ˇʽ ʰ˄ʰ˂ˏˋʶʽˌ ʵʽʶ˄ʶˊʴʺʻʹˁʰ˄ ˋˏ˃˒˖˄ʰ ˃ʶ ˍʽˌ ʃˊˈˍˎˉʶˌ 
ɾʶʻˈʵˇˎˌ ώммϐΣ ʶ˄˗ ʴʽʰ ˍʹ ˃ʷˍˊʹˋʹ ˍʹˌ ˉʶˊʽʶˁˍʽˁˈˍʹˍʰˌ ˋʶ ˃ʶʻʱ˄ʽˇ ˋˍˇ ˉʰˊʰʴˈ˃ʶ˄ˇ ʲʽˇʰʷˊʽˇ 
˔ˊʹˋʽ˃ˇˉˇʽʺʻʹˁʶ ʰʷˊʽˇˌ ˔ˊ˖˃ʰˍˇʴˊʱ˒ˇˌ ό {ILa!5½¦ D/πнлмл ǇƭǳǎύΦ 

 

 

ɳʽˁˈ˄ʰ м ʃʶʽˊʰ˃ʰˍʽˁʺ ʵʽʱˍʰ˅ʹ PABR 



ʃʾ˄ʰˁʰˌ 1 ʃʰˊʱ˃ʶˍˊˇʽ ˂ʶʽˍˇˎˊʴʾʰˌ ʴʽʰ ˁʱʻʶ ˉʶʽˊʰ˃ʰˍʽˁʺ ˒ʱˋʹ 

ɽʶʽˍˇˎˊʴʽˁʷˌ ʃʰˊʱ˃ʶˍˊˇʽ ɳʴˁ˂ʽ˃ʰˍʽˋ˃ˈˌ ʃʶʽˊʰ˃ʰˍʽˁʺ ˒ʱˋʹ 

ʇʵˊʰˎ˂ʽˁˈˌ ʋˊˈ˄ˇˌ ʃʰˊʰ˃ˇ˄ʺˌ н ɾʷˊʶˌ м ɾʷˊʰ 

ʄˎʻ˃ˈˌ ʁˊʴʰ˄ʽˁʺˌ ʊˈˊˍʽˋʹˌ лΣфу Ǝ/h5κό[ϝŘύ лΣфт Ǝ/h5κό[ϝŘύ 

ʃʶˊʾˇʵˇˌ ɳ˄ʰ˂˂ʰʴʺˌ ɲʽʰ˃κˍ˖˄  π π 

ɸʶˊ˃ˇˁˊʰˋʾʰ ɽʶʽˍˇˎˊʴʾʰˌ орƻ/ орƻ/ 

 

 

3. ɮʃʁʆɳɽɳʅɾɮʆɮ ɼɮɹ ʅʇɵɶʆɶʅɶ 

 

ʅˍˇ˄ ʃʾ˄ʰˁʰ н ˉʰˊˇˎˋʽʱʸˇ˄ˍʰʽ ˍʰ ˋˎʴˁʶ˄ˍˊ˖ˍʽˁʱ ʰˉˇˍʶ˂ʷˋ˃ʰˍʰ ˍʹˌ ˒ʱˋʹˌ ʶʴˁ˂ʽ˃ʰˍʽˋ˃ˇˏ ˁʰʽ ˍʹˌ 
ˉʶʽˊʰ˃ʰˍʽˁʺˌ ˒ʱˋʹˌ ˂ʶʽˍˇˎˊʴʾʰˌΦ ɽˈʴ˖ ˍʹˌ ˔ʰ˃ʹ˂ʺˌ ˋˎʴˁʷ˄ˍˊ˖ˋʹˌ ˋʶ /h5 ̱ ˇˎ ˍʶ˂ʽˁˇˏ ʵʽʰ˂ˏ˃ʰˍˇˌ 
ʰ˄ʱ˃ʶʽ˅ʹˌ ˍ˖˄ ʵˏˇ ʰˉˇʲ˂ʺˍ˖˄Σ ˇ Iw¢ ̱ ˇˎ ʰ˄ˍʽʵˊʰˋˍʺˊʰ ˃ʶʽ˗ʻʹˁʶ ˋˍʹ˄ ˃ʾʰ ˃ʷˊʰ ˃ʶ ˋˁˇˉˈ ʹ 
ˉʶʽˊʰ˃ʰˍʽˁʺ ˒ʱˋʹ ˂ʶʽˍˇˎˊʴʾʰˌ ˄ʰ ʷ˔ʶʽ ˍˇ˄ ʾʵʽˇ ˊˎʻ˃ˈ ˇˊʴʰ˄ʽˁʺˌ ˒ˈˊˍʽˋʹˌ ˃ʶ ˍʹ˄ ˒ʱˋʹ ʶʴˁ˂ʽ˃ʰˍʽˋ˃ˇˏΦ 
ʂˋˇ˄ ʰ˒ˇˊʱ ˍʹ˄ ˉʰˊʰʴ˖ʴʺ ʲʽˇʰʶˊʾˇˎ ˁʰʽ ʲʽˇ˃ʶʻʰ˄ʾˇˎ ˇ ʰ˄ˍʽʵˊʰˋˍʺˊʰˌ ʵʶ˄ ˒ʰʾ˄ʶˍʰʽ ˄ʰ ʶˉʹˊʶʱʸʶˍʰʽ 
ˋʹ˃ʰ˄ˍʽˁʱ ʰˉˈ ˍʹ˄ ˉˊˇˋʻʺˁʹ ˍˇˎ ŎƻƴŘŜƴǎŀǘŜ ˁʰʽ ˍʹ˄ ˃ʶʾ˖ˋʹ ˍˇˎ Iw¢ ̍ˉ˖ˌ ˒ʰʾ˄ʶˍʰʽ ˋˍˇ όɲʽʱʴˊʰ˃˃ʰ мύΦ 
ɳ˄ʵʽʰ˒ʷˊˇ˄ ˉʰˊˇˎˋʽʱʸʶʽ ʹ ˋˏʴˁˊʽˋʹ ˍˇˎ ˉʰˊʰʴˈ˃ʶ˄ˇˎ ˃ʶʻʰ˄ʾˇˎ ˃ʶ ˍˇ ʻʶ˖ˊʹˍʽˁˈ ˃ʷʴʽˋˍˇ ˍʹˌ 
ʵʽʶˊʴʰˋʾʰˌ ˉˇˎ ʶ˅ʰˊˍʱˍʰʽ ʰˉˈ ˍʹ˄ ˁʰˍʰ˄ʱ˂˖ˋʹ ˇˊʴʰ˄ʽˁˇˏ ˒ˇˊˍʾˇˎ όɲʽʱʴˊʰ˃˃ʰ нύΦ ʅˍˇ ɲʽʱʴˊʰ˃˃ʰ н 
ˉʰˊʰˍʹˊʶʾˍʰʽ ˈˍʽ ˇ ʰ˄ˍʽʵˊʰˋˍʺˊʰˌ ˉʰˊˇˎˋʽʱʸʶʽ ˔ʰ˃ʹ˂ʷˌ ʰˉˇʵˈˋʶʽˌ ˁʰˍʱ ˍʹ˄ ˉˊ˗ˍʹ ˒ʱˋʹ ˂ʶʽˍˇˎˊʴʾʰˌ 
ʶ˄˗ ˋˍʹ˄ ˋˎ˄ʷ˔ʶʽʰ ˉʰˊˇˎˋʽʱʸʶʽ ʽˁʰ˄ˇˉˇʽʹˍʽˁʺ ʰˉˈʵˇˋʹ ʴʽʰ ˍˇ˄ ˊˎʻ˃ˈ ˇˊʴʰ˄ʽˁʺˌ ˒ˈˊˍʽˋʹˌ ˋˍˇ˄ ˇˉˇʾˇ 
˂ʶʽˍˇˎˊʴʶʾΦ ʂˋˇ˄ ʰ˒ˇˊʱ ˋˍʹ˄ ˁʰˍʰ˄ʱ˂˖ˋʹ ˍˇˎ ˇˊʴʰ˄ʽˁˇˏ ˒ˇˊˍʾˇˎΣ ʹ ˉʶʽˊʰ˃ʰˍʽˁʺ ˒ʱˋʹ ˂ʶʽˍˇˎˊʴʾʰˌ 
ˉʰˊˇˎˋʽʱʸʶʽ ʰˊˁʶˍʱ ʽˁʰ˄ˇˉˇʽʹˍʽˁˈ ʲʰʻ˃ˈ ʰˉˈʵˇˋʹˌΦ ɳˉʾˋʹˌ ʶˉʽˍˎʴ˔ʱ˄ʶˍʰʽ ˋʹ˃ʰ˄ˍʽˁʺ ˃ʶʾ˖ˋʹ ˍˇˎ 
ˇˊʴʰ˄ʽˁˇˏ ˒ˇˊˍʾˇˎ ˍˇˎ ˃ʾʴ˃ʰˍˇˌ ˃ʷˋ˖ ˍʹˌ ʰ˄ʰʶˊˈʲʽʰˌ ˔˗˄ʶˎˋʹˌΣ ˈˉ˖ˌ ˒ʰʾ˄ʶˍʰʽ ˋˍˇ όɲʽʱʴˊʰ˃˃ʰ оύΦ 

 

ʃʾ˄ʰˁʰˌ н ʅˎʴˁʶ˄ˍˊ˖ˍʽˁʱ ʰˉˇˍʶ˂ʷˋ˃ʰˍʰ ˂ʶʽˍˇˎˊʴʾʰˌ ˍˇˎ ʰ˄ˍʽʵˊʰˋˍʺˊʰΦ 

 ɳʴˁ˂ʽ˃ʰˍʽˋ˃ˈˌ ʃʶʽˊʰ˃ʰˍʽˁʺ ˒ʱˋʹ 

ɮˉˇˍʶ˂ʷˋ˃ʰˍʰ ɾʷˋʹ ˍʽ˃ʺ ɾʷˋʹ ˍʽ˃ʺ 

ʃʰˊʰʴ˖ʴʺ ɰʽˇʰʶˊʾˇˎ ό[κŘύ оуΣт ҕ пΣп опΣр ҕ оΣн 

ʃʰˊʰʴ˖ʴʺ ɾʶʻʰ˄ʾˇˎ ό[κŘύ нмΣм ҕ оΣр нлΣс ҕ оΣо 

ʃˇˋˇˋˍˈ ɾʶʻʰ˄ʾˇˎ ό҈ύ рсΣн ҕ сΣф соΣм ҕ тΣр 

/h5 ˎˉˇˋˍˊ˗˃ʰˍˇˌ όƎκ[ύ мΦфу ҕ лΦн лΦфтҕ лΦлу 



ɼʰˍʰ˄ʱ˂˖ˋʹ /h5 ό҈ύ ту ҕ мтΣт фмҕ пΣр 

ɮˉˈʵˇˋʹ ˖ˌ ˉˊˇˌ /Iп ό҈ύ ро ҕ мфΣт уу ҕ мсΣо 

 

 



ɲʽʱʴˊʰ˃˃ʰ 1 ʃʰˊʰʴ˖ʴʺ ɰʽˇʰʶˊʾˇˎ ˁʰʽ ʲʽˇ˃ʶʻʰ˄ʾˇˎ ʰ˄ʱ ʹ˃ʷˊʰ ˂ʶʽˍˇˎˊʴʾʰˌΦ 

 

ɲʽʱʴˊʰ˃˃ʰ 2 ʃʰˊʰʴˈ˃ʶ˄ˇ ˃ʶʻʱ˄ʽˇ ˁʰʽ ʰ˄ʰ˃ʶ˄ˈ˃ʶ˄ˇ ʻʶ˖ˊʹˍʽˁˈ ʰ˄ʱ ʹ˃ʷˊʰ ˂ʶʽˍˇˎˊʴʾʰˌΦ 

 

ɲʽʱʴˊʰ˃˃ʰ 3 ɲʽʰ˂ˎˍˈ COD ̀ ˍʹ˄ ˍˊˇ˒ˇʵˇˋʾʰ ˁʰʽ ˋˍʹ˄ ʶˁˊˇʺ ˍˇˎ ʰ˄ˍʽʵˊʰˋˍʺˊʰΦ 

 

ʆˇ ˋˎʴˁʶˁˊʽ˃ʷ˄ˇ ʰˉˇˍʷ˂ʶˋ˃ʰ ʶʾ˄ʰʽ ʶ˄ʻʰˊˊˎ˄ˍʽˁˈ ˁʰʻ˗ˌ ʵʶʾ˔˄ʶʽ ˈˍʽ ʹ ˋˎʴˁʶˁˊʽ˃ʷ˄ʹ ˃ʶʻˇʵˇ˂ˇʴʾʰ ʻʰ 
˃ˉˇˊˇˏˋʶ ˄ʰ ʰ˄ˍʽˁʰˍʰˋˍʺˋʶʽ ˍʹ˄ ʶ˄ʶˊʴʶʽʰˁʱ ʰˋˏ˃˒ˇˊʹ ʵʽʶˊʴʰˋʾʰ ˍʹˌ ʲʽˇ˂ˇʴʽˁʺˌ ˇ˅ʶʾʵ˖ˋʹˌ ʰ˄ ˈ˔ʽ ʴʽʰ 
ˍʹ˄ ˉ˂ʺˊʹ ʴˊʰ˃˃ʺ ʶˉʶ˅ʶˊʴʰˋʾʰˌ ˍ˖˄ ʰˋˍʽˁ˗˄ ˂ˎ˃ʱˍ˖˄Σ ʴʽʰ ʷ˄ʰ ˉˇˋˇˋˍˈ ˍʹˌΦ ʅˎʴˁʶˁˊʽ˃ʷ˄ʰ ˉʰˊʰˍʹˊʶʾˍʰʽ 
ˈˍʽ ˃ʶ ˍʹ˄ ˉʱˊˇʵˇ ˍʹˌ ɻʽʶˊʴʰˋʾʰˌ  ʹ ˁʰˍʰ˄ʱ˂˖ˋʹ ˇˊʴʰ˄ʽˁˇˏ ˒ˇˊˍʾˇˎ ˅ʶˉʶˊ˄ʱʶʽ ˍˇ фл҈Φ ʆˇ ʴʶʴˇ˄ˈˌ ʰˎˍˈ 
ˋʶ ˋˎ˄ʵˎʰˋ˃ˈ ˃ʶ ˍʹ˄ ˍʰ˔ˏˊˊˎʻ˃ʹ ˂ʶʽˍˇˎˊʴʾʰ ˍˇˎ ʰ˄ˍʽʵˊʰˋˍʺˊʰ ʵʹ˂ʰʵʺ ˍʹ˄ ʽˁʰ˄ˈˍʹˍʰ ˍˇˎ ˄ʰ 
ʶˉʶ˅ʶˊʴʱʸʶˍʰʽ ˃ʶʴʱ˂ˇˎˌ ˈʴˁˇˎˌ ʰˉˇʲ˂ʺˍ˖˄ ʹ˃ʶˊʹˋʾ˖ˌ ˁʰʻʽˋˍʱ ˍʹ˄ ʵʽʶˊʴʰˋʾʰ ʲʽ˗ˋʽ˃ʹΦ  

 

 



ɳʇʋɮʄɹʅʆɹɳʅ 

ɶ ʁ ˊʶˎ˄ʹˍʽˁʺ ʁ ˊʴʰˋʾʰ ˎ ˉˇˋˍʹˊʾ˔ˍʹˁʶ h ˉˈ ̱  ̌ɳ˂˂ʹ˄ʽˁˈ ɺʵˊˎ˃ʰ ɴˊʶˎ˄ʰˌ ˁʰʽ 
ɼʰʽ˄ˇˍˇ˃ʾʰˌ όɳɽΦɹɲΦɳΦɼΦύ ˋˍˇ ˉ˂ʰʾˋʽˇ ˍʹˌ ɲˊʱˋʹˌ ζмʹ ʃˊˇˁʺˊˎ˅ʹ ʶˊʶˎ˄ʹˍʽˁ˗˄ 
ʷˊʴ˖˄ ɳɽΦɹɲΦɳΦɼΦ ʴʽʰ ˍʹ˄ ʶ˄ʾˋ˔ˎˋʹ ˍ˖˄ ˃ʶ˂˗˄ ɲɳʃ ˁʰʽ ɳˊʶˎ˄ʹˍ˗˄κˍˊʽ˗˄ ˁ ʰʽ ̱ ʹ˄ 
ˉˊˇ˃ʺʻʶʽʰ ʁ ˊʶˎ˄ʹˍʽˁˇˏ ʁ ˅ˇˉ˂ʽˋ˃ˇˏ ˃ ʶʴʱ˂ʹˌ ʰ˅ʾʰˌη όɮˊʽʻ˃ˈˌ ɴˊʴˇˎΥ нтфтύΦ 
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